## 1. INTRODUCTION

- 1. The International Accounting Standards Committee (IASC) is an independent private sector body formed in 1973 through an agreement made by professional accountancy bodies from Australia, Canada, France, Germany, Japan, Mexico, the Netherlands, the United Kingdom and Ireland and the United States of America. Since 1983, IASC's Members have been all the professional accountancy bodies that are Members of the International Federation of Accountants (IFAC). As at 1 November 1998, IASC and IFAC have 143 Members in 103 countries.
- 2. IASC was founded to formulate and publish, in the public interest, International Accounting Standards to be observed in the presentation of published financial statements and to promote their worldwide acceptance and observance. It was envisaged that International Accounting Standards should be capable of worldwide acceptance and contribute to a significant improvement in the quality and comparability of corporate disclosure. This vision has not changed since 1973 and there is no reason why it should change in the foreseeable future.
- 3. IASC has issued over 30 International Accounting Standards. These standards deal with topics that affect the financial statements of business enterprises. The Board has also issued a Framework for the Preparation and Presentation of Financial Statements, which helps the Board to achieve conceptual consistency within International Accounting Standards. The Framework states that its purpose is, among other things, to assist the Board in:
  - (a) developing new International Accounting Standards and in reviewing existing International Accounting Standards; and
  - (b) promoting the harmonisation of regulations, accounting standards and procedures relating to the presentation of financial statements by providing a basis for reducing the number of alternative accounting treatments permitted by International Accounting Standards.
- 4. International Accounting Standards have done a great deal both to improve and to harmonise financial reporting around the world. They are used:
  - (a) as a basis for national accounting requirements in many countries;
  - (b) as an international benchmark by some countries that develop their own requirements (including certain major industrialised countries, regional organisations such as the European Union, and an increasing number of emerging markets such as China and many other countries in Asia, Central Europe and the countries of the former Soviet Union);
  - (c) by stock exchanges and regulatory authorities that allow foreign or domestic companies to present financial statements in accordance with International Accounting Standards;
  - (d) by supra-national bodies that rely on IASC to produce accounting standards that improve the quality of financial reporting and the comparability of financial statements, instead of developing their own requirements;

- (e) by the World Bank Group and other development agencies that require borrowers and recipients of other forms of aid to follow high standards of financial reporting and accountability; and
- (f) by a growing number of individual companies.
- 5. IASC has been successful in developing high quality International Accounting Standards that have gained increasing acceptance around the world. To build on that achievement, the Working Party believes that IASC should now be anticipating future needs and modifying its own role and structure in response to major changes in the environment in which IASC operates. Some of these changes have been driven by external factors, while others have been driven partly by IASC's own achievements.
- 6. The changing role of IASC creates new challenges and opportunities for IASC. Although IASC's existing structure has served it well, the Working Party believes that IASC needs to change its structure so that it can meet these new challenges, and seize those opportunities, as effectively as it has responded to other challenges and opportunities in the first 25 years of its life.
- 7. Section 2 of this Paper discusses the changes in IASC's environment and the new challenges that these changes create for IASC. Section 3 reviews IASC's objectives and section 4 summarises IASC's current structure and due process. Section 5 develops a model for strengthening IASC's structure and due process to keep pace with the changes in the environment. Section 6 deals with implementation, enforcement and training issues. Last, and far from least, section 7 looks at how IASC's funding can be improved.

## 2. NEW CHALLENGES FOR IASC

- 8. For the first 25 years of its existence, IASC has mainly been a harmoniser a body that selects an accounting treatment that already exists at the national level in some countries and then seeks worldwide acceptance of that treatment, perhaps with some modifications. IASC's current structure has enabled IASC to play an effective role as a harmoniser.
- 9. In recent years, IASC has taken on a more innovatory role in certain areas. For example, in March 1997, IASC's Steering Committee on Financial Instruments published proposals for dealing with financial assets and financial liabilities and those proposals went beyond any requirements or proposals that existed at a national level.
- 10. The pace of change in business life and the financial world is accelerating rapidly. The Working Party believes that in the future IASC will need to be an innovator and an initiator to a much greater extent than it is today. IASC can meet this challenge only by working closely with national standard setters.
- 11. Another increasingly important challenge for IASC will be to work with national standard setters to bring about convergence between national accounting standards and International Accounting Standards around solutions requiring listed enterprises (i.e. enterprises with publicly traded equity or debt securities) and other economically significant enterprises to report high-quality, transparent and comparable information that will help participants in capital markets and others to make economic decisions. The standards of many countries are already converging with International Accounting Standards. However, trends such as globalisation and the increasing pace of business and financial change have made this task more urgent. The Working Party believes that IASC and national standard setters need to find new ways of working together to minimise unnecessary delays in reaching consensus and implementing the results of that consensus.

#### **CHANGES IN IASC'S ENVIRONMENT**

- 12. In the Working Party's view, the new challenges facing IASC result from significant changes in IASC's environment in recent years. The most important of these changes are:
  - (a) a rapid growth in international capital markets, combined with an increase in cross-border listings and cross-border investment. These have led to efforts by securities regulators to develop a common "passport" for cross-border securities listings and to achieve greater comparability in financial reporting;
  - (b) efforts of global organisations (such as the World Trade Organisation) and regional bodies (such as the European Union, NAFTA, MERCOSUR and APEC) to dismantle barriers to international world trade;
  - (c) a trend towards the internationalisation of business regulation;
  - (d) increasing influence of International Accounting Standards on national accounting requirements and practice;
  - (e) accelerating innovation in business transactions;

- (f) increasing demand from users for new types of financial and other performance information;
- (g) new developments in the electronic distribution of financial and other performance information; and
- (h) growing need for relevant and reliable financial and other performance information both in countries in transition from planned economies to market economies and in developing and newly industrialised economies.
- 13. These changes are discussed below. They have placed strain on the organisational and financial resources of a body that relies, as IASC does, extensively on work by volunteers and on relatively informal contacts with national standard setters.

## **International Capital Markets**

14. In recent years, there has been explosive growth in the provision of international capital, in the opportunities for international investment and in the number and size of international securities offerings, as shown by the table set out below. This has fuelled a demand for a core set of high-quality International Accounting Standards that business enterprises from all countries can use when they seek access to the international capital markets. This has brought an increase in cross-border listings and cross-border investment. It has also led to demand for greater comparability in accounting and disclosure practices.

Cross-Border Transactions in Bonds and Equities As a Percentage of Gross Domestic Product (GDP) (Gross purchases and sales of securities between residents and non-residents)

,	1975	1980	1985	1990	1995	1997
United States	4	9	35	89	135	213
Japan	2	8	62	119	65	96
Germany	5	7	33	57	172	253
France	-	5	21	54	187	313
Italy	1	1	4	27	253	672
Canada	3	9	27	65	189	358

Source: Table V.1, Bank for International Settlements, 68th Annual Report (1998)

- 15. The acceptance by stock exchanges of International Accounting Standards brings great benefits to users of financial statements and to business enterprises that report under International Accounting Standards:
  - (a) users benefit by having high-quality financial information that is comparable from one enterprise to another, comprehensive and readily comprehensible; and
  - (b) enterprises benefit by having access to a larger pool of foreign capital at a lower cost and by saving the cost of preparing reconciliation statements to another set of accounting standards for cross-border listings. Also, the increasing convergence of national standards (through convergence with International Accounting Standards) reduces the need to restate financial information for consolidation purposes.

- 16. Many stock exchanges accept International Accounting Standards for cross-border listing purposes, but stock exchanges in Canada, Japan and the United States are currently exceptions. IASC has therefore adopted a work programme to produce a comprehensive core set of high quality standards, aiming at more general acceptance of International Accounting Standards for cross-border listings. The Technical Committee of the International Organization of Securities Commissions (IOSCO) has agreed that when IASC completes the work programme to the Technical Committee's satisfaction, the Technical Committee will consider recommending that IOSCO should endorse International Accounting Standards for cross-border capital raising and listing purposes in all global markets. IASC aims to complete the core standards in 1998.
- 17. If IOSCO endorses International Accounting Standards (and International Standards on Auditing), there is likely to be a substantial increase in the credibility of financial statements prepared under International Accounting Standards and audited under International Standards on Auditing.
- 18. Some commentators believe that capital markets are moving towards becoming not national capital markets but part of a single global capital market. Illustrations of this development are:
  - (a) the development of international securities trading over the Internet;
  - (b) recent cross-border alliances between national stock exchanges; and
  - (c) the creation of EASDAQ, a pan-European over-the-counter market.

These commentators believe that movement towards a single global capital market will inevitably lead to demands for securities and stock market regulation to move from the national arena to the international arena and for a similar development in the process for setting accounting standards.

19. At present, enterprises with cross-border listings sometimes have to report under two different accounting regimes. In a number of well-publicised cases, such enterprises have reported substantially different information under the two regimes. This has caused confusion for users of that financial information. The reporting of such significant differences may cause even greater confusion as the circle of cross-border investors becomes wider.

## **World Trade**

- 20. Global and regional organisations and initiatives (such as the World Trade Organisation, the OECD, the European Union, NAFTA, MERCOSUR and APEC) are making great efforts to liberalise international trade. These efforts have stimulated international trade and also added impetus to the trend of globalisation that is apparent in many sectors. The result is a growing demand for business enterprises to present relevant, reliable, comparable and comprehensible financial information to users in other countries. Several of these bodies are looking to IASC to produce standards that will satisfy this demand. For example:
  - (a) at the World Trade Organisation conference in 1996, government ministers issued a statement in 1996 encouraging "the successful completion of international standards in the accountancy sector by IASC, IFAC and IOSCO";

- (b) the European Commission announced in 1995 that, rather than create a European Accounting Standards Board or a new layer of European Accounting Standards, it would associate the European Union with the efforts undertaken by IASC and IOSCO towards a broader international harmonisation of accounting standards. The European Commission plays a leading role in coordinating the development of financial reporting in the European Union; and
- (c) the Business Forum of the Asia Pacific Economic Co-operation (APEC) countries gave support in 1996 to the internationalisation of accounting through a core of consistent high quality International Accounting Standards that would be acceptable for general purpose accounting guidelines in all APEC countries.
- 21. Convergence of national accounting standards will make it easier for public accountants to practise across national borders. This will add to competition in accountancy services and may reduce their cost.

## **Internationalisation of Business Regulation**

22. Many business activities are subject to supervision by government regulatory agencies. Increasingly, these agencies are looking to develop co-ordinated international approaches to supervision through bodies such as IOSCO, the Basle Committee on Banking Supervision and the International Association of Insurance Supervisors. International co-operation on supervisory matters is likely to increase in response to the recent turmoil in many financial markets. In a number of cases, regulatory and supervisory requirements rely on information in financial statements. Such requirements are likely to be more effective if they are based on consistent application of uniform accounting requirements.

## **National Accounting Standards**

- 23. The prospect of IOSCO endorsement and the growing international support for IASC have given national standard setters increasing incentives to bring their own requirements into line with International Accounting Standards. As a result, many countries have already adopted International Accounting Standards or are in the process of bringing their domestic standards into line with International Accounting Standards. In addition, many companies from countries whose accounting standards are not widely recognised have chosen to adopt International Accounting Standards. This trend has raised IASC's profile and increased the interest in its work. IASC's work now has a great impact on many more groups, in many more countries, than before. These groups now want to participate in IASC's decisions.
- 24. Several national standard setters have stated publicly that they accept the need for, and are committed to, international harmonisation. For example:
  - (a) in their Policy Statement 6, International Harmonisation Policy, the Australian Accounting Standards Board (AASB) and the Public Sector Accounting Standards Board (PSASB) state that their objective is to pursue the development of an internationally accepted set of accounting standards which can be adopted in Australia. Their interim objective is to work towards ensuring that compliance with Australian Accounting Standards results in compliance with International Accounting Standards;

- (b) a Task Force on Standard Setting set up by the Canadian Institute of Chartered Accountants reported in May 1998 that "the internationalization of standard setting for private sector accounting and auditing is inevitable". One long term goal set out in the Task Force's report was as follows. "There will be one set of internationally-accepted accounting standards in the private sector. Canada will play a significant role in establishing international accounting standards and retain its authority to set unique Canadian Accounting Standards where circumstances warrant":
- (c) the Chairman of the UK's Accounting Standards Board wrote in his Progress Report for 1997 that "the Board has adopted a policy that combines a sincere commitment to work for international harmonisation with the flexibility to ensure that each issue is evaluated with full regard to the needs of financial reporting in the UK and the Republic of Ireland"; and
- (d) one objective set out in the strategic plan of the USA's Financial Accounting Standards Board (FASB) is to promote the development and acceptance of a superior set of international accounting standards.
- 25. An important recent development in Europe is new legislation in Belgium, France, Germany and Italy that allows some domestic companies to use International Accounting Standards instead of domestic standards in their consolidated financial statements, subject to certain conditions. Other countries in the European Union are considering similar measures.
- 26. These European countries will still permit or require some enterprises to use existing domestic requirements instead of International Accounting Standards. In time, the desire for comparability within individual countries may create pressure for those domestic requirements to converge towards International Accounting Standards.
- 27. In the National Securities Markets Improvements Act of 1996, the US Congress stated that the "establishment of a high-quality comprehensive set of generally accepted international accounting standards would greatly facilitate international financing activities and, most significantly, would enhance the ability of foreign corporations to access and list in the United States markets". It called on the Securities and Exchange Commission (SEC) to "enhance its vigorous support for the development of high-quality international accounting standards as soon as practicable".

#### **Innovation in Business Transactions**

28. Business and finance are becoming more complex with the result that accounting standards have had to deal with more complex issues. Prominent examples are the growth of trading in derivatives and the increasing use of securitisation transactions. IASC has dealt with this increased complexity by, for example, improving its research before developing new and revised Standards, consulting specialists, improving its due process, and sharing resources with national standard setting bodies. The complexities and controversies will undoubtedly continue and IASC must make sure that it is in a position to address them effectively.

## **Demand for New Types of Financial and Other Performance Information**

29. Increasingly, users of financial statements are calling for information that is not easily captured by traditional accounting techniques. For example, as the service sector becomes more prominent in many economies, users are starting to ask for information about human resources and intangible assets. Similarly, growing concern for environmental issues is leading to requests for more comprehensive reporting in this area. IASC will need to draw on greater research and innovation capability (either its own or that of national standard setters) to deal with these emerging topics effectively.

## **Electronic Distribution of Information**

30. Enterprises are beginning to use channels such as the Internet and CD-ROMs to distribute financial and other performance information more quickly and in greater volume. The availability of greater computing power is also making it feasible to generate information of a kind or quality that was not available only a few years ago. In time, enterprises may be reporting on a real-time basis. This trend may call for changes in the nature of financial reporting standards. Also, standard setters may need to find new mechanisms for responding quickly to new reporting practices stemming from the rapid innovation in information technology.

## **Countries in Transition and Developing and Newly Industrialised Countries**

- 31. Several countries in East and Central Europe and in Asia are in transition from centrally planned economies to market driven economies. This change has led to a need for new national accounting requirements. Many of these countries have looked, or are looking, to International Accounting Standards as the basis for their new national accounting standards.
- 32. Many developing and newly industrialised countries are using International Accounting Standards as their national requirements, or as the basis for their national requirements. These countries have a growing need for relevant and reliable financial information to meet the requirements both of domestic users and of international providers of the capital that they need.
- 33. International Accounting Standards help borrowers and recipients of aid to meet the strict accountability required by the World Bank and other development and aid agencies.
- 34. IASC liaises closely with intergovernmental bodies that seek to improve accounting practices in countries in transition and in developing and newly industrialising countries. These include:
  - (a) the World Bank Group;
  - (b) the United Nations Conference on Trade and Development (UNCTAD) and UNCTAD's Intergovernmental Working Group of Experts on International Standards of Accounting and Reporting (ISAR); and
  - (c) the Organisation for Economic Co-operation and Development (OECD).

#### IMPACT OF THESE ENVIRONMENTAL CHANGES ON IASC

- 35. The trends discussed above show a clear and growing demand from the market for high-quality global accounting standards that provide transparency and comparability. Indeed, in October 1998
  - (a) the G-22 Working Party on Transparency and Accountability reported that: "weaknesses in the provision and use of information played a major part in the development and spread of recent international financial crises." The report called for "a set of high quality, internationally acceptable accounting standards"; and
  - (b) a declaration of G7 Finance Ministers and Central Bank Governors on 30 October stated, among other things: "We call upon (...) the IASC to finalise by early 1999 a proposal for a full range of internationally agreed accounting standards. IOSCO, IAIS, and the Basle Committee should complete a timely review of these standards. (...) We commit ourselves to endeavour to ensure that private sector institutions in our countries comply with these principles, standards and codes of best practice. We call upon (...) all countries which participate in global capital markets similarly to commit to comply with these internationally agreed codes and standards (...)"

Similarly, the Chairman of the Basle Committee on Banking Supervision has stated that "the Basle Committee considers transparency to be a key element of an effectively supervised, safe and sound banking system".

- 36. IASC's international structure and record of success have put it in a unique position to satisfy the demand for high-quality global accounting standards. However, IASC cannot take further success for granted. Among other things, IASC's role in the future is unlikely to be the same as in the past. In its early years, IASC acted mainly as a consolidator of existing national standards. In more recent times, it has begun to combine that role with the role of a catalyst a co-ordinator of national initiatives and an initiator of new work at the national level. In the future, IASC's role as a catalyst and initiator should become more prominent.
- 37. Another important consideration is the vastly increased significance of IASC's work. IASC's structure worked well when IASC's work affected a relatively small number of countries and enterprises. There is no guarantee that this structure will work without modification at a time when IASC's work has a direct or indirect effect in almost every country.
- 38. As explained above, innovation in business transactions is accelerating, demand from users for new types of financial and other performance information is increasing and there are rapid developments in electronic distribution of information. Also, the life cycle of standards in all fields not just in accounting is shrinking rapidly. IASC needs a structure that will enable it to cope effectively with these and other new developments.

- 39. Since the beginning of 1997, IASC has achieved impressive results by increasing the length and frequency of Board meetings from three meetings a year of three or four days each to four meetings a year of five, six or even seven days each. However, it would be difficult for volunteers to maintain this level of activity indefinitely rather than for a limited time to achieve a well-specified and valuable objective. The status quo, as it has been since the beginning of 1997 is not sustainable. The choice is between the status quo as it was before 1997 and something else.
- 40. In the Working Party's view, IASC must now consider structural changes so that it can continue to meet the need for high-quality global accounting standards. If IASC fails to meet that need, other national, regional or international bodies are likely to emerge to fill the gap in response to market pressures and become de facto global or regional standard setters.

## 3. OBJECTIVES OF IASC

- 41. The objectives of IASC as stated in its Constitution are:
  - (a) to formulate and publish in the public interest accounting standards to be observed in the presentation of financial statements and to promote their worldwide acceptance and observance; and
  - (b) to work generally for the improvement and harmonisation of regulations, accounting standards and procedures relating to the presentation of financial statements.
- 42. In the Working Party's view, it is important to focus IASC's objectives more precisely as follows:
  - (a) to develop International Accounting Standards that require high-quality, transparent and comparable information which will help participants in capital markets and others to make economic decisions; and
  - (b) to promote the use of International Accounting Standards by working with national standard setters to:
    - (i) bring about convergence, for listed enterprises (i.e. enterprises with publicly traded equity or debt securities) and other economically significant enterprises, between national accounting standards and International Accounting Standards; and
    - (ii) encourage national, regional and international authorities to permit or require unlisted enterprises that, individually, are not economically significant to use International Accounting Standards if those Standards meet the needs of the users of the financial statements of such enterprises.
- 43. The following aspects of this conclusion are discussed below:
  - (a) the need for high-quality, transparent and comparable information to support economic decisions;
  - (b) working with national standard setters;
  - (c) convergence of standards for listed and other economically significant enterprises;
  - (d) small and medium-sized enterprises (SMEs);
  - (e) countries in transition and developing and newly industrialised countries; and
  - (f) the public sector and not-for-profit organisations.

# HIGH-QUALITY, TRANSPARENT AND COMPARABLE INFORMATION TO SUPPORT ECONOMIC DECISIONS

- 44. The Working Party believes that it is vital for IASC to continue to use an agreed conceptual Framework (the Framework for the Preparation and Presentation of Financial Statements) to ensure that its standards are of high quality and require transparent and comparable information to help participants in capital markets and others to make economic decisions. The Framework may need to be revised from time to time on the basis of IASC's experience of working with it.
- 45. IASC's Framework states that it is concerned with general purpose financial statements directed toward the common information needs of a wide range of users. The Framework argues that the objective of financial statements is to provide information about the financial position, performance and changes in financial position of an enterprise that is useful to a wide range of users in making economic decisions.
- 46. If financial statements meet the needs of participants in the capital markets (present and potential investors and lenders), the Framework suggests they will also meet most of the needs that are common to other users of financial statements, such as employees, suppliers and other trade creditors, governments and their agencies and the public.
- 47. The Working Party strongly supports the Framework's focus on information that will meet the needs of the capital markets and so also meet most of the common needs of other users.
- 48. The Framework identifies four qualitative characteristics that make the information in financial statements useful to users. In summary, the information should be:
  - (a) readily understandable by users;
  - (b) relevant to their decision-making needs;
  - (c) reliable; and
  - (d) comparable with information provided by the enterprise itself in its financial statements through time and with information provided in the financial statements of different enterprises.
- 49. The Framework states that information is relevant to the decision-making needs of users when it helps them to evaluate past, present or future events or confirm, or correct, their past evaluations. For example, information about the current financial position and past performance and cash flows has value to users when they evaluate the ability of an enterprise to generate cash and cash equivalents.
- 50. To be reliable, the Framework argues that information must:
  - (a) represent faithfully the transactions and other events it either purports to represent or could reasonably be expected to represent;

- (b) account for, and present, transactions and other events that it purports to represent in accordance with their substance and economic reality and not merely their legal form;
- (c) be neutral, that is free from bias;
- (d) contend with the uncertainties that inevitably surround many events and circumstances by the exercise of prudence; and
- (e) be complete within the bounds of materiality and cost.
- 51. The Framework notes the need for a balancing, or trade-off, between the four qualitative characteristics listed in paragraph 48 above. It also recognises that the provision of relevant and reliable information may be constrained by the need for timely reporting and for a balance between costs and benefits.
- 52. The Working Party believes that the qualitative characteristics identified by the Framework are a sound basis for judging whether information is of high quality and will help users to make economic decisions.
- 53. When IASC completes the core standards, it is likely that IASC will have to give some attention to issues such as:
  - (a) a review of IASC's Framework for the Preparation and Presentation of Financial Statements;
  - (b) care, maintenance and possible amendment of existing IASC standards, particularly those approved before the Framework was developed; and
  - (c) industry standards. IASC has issued a standard, IAS 30, for banks and similar financial institutions, and has started projects on agriculture, insurance and the extractive industries (mining, oil and gas). In due course, IASC may consider projects on other industries.
- 54. In creating International Accounting Standards, IASC has generally adopted one or more accounting treatments that already existed at the national level in some countries, perhaps with some modifications in certain cases. In recent years, IASC has taken on a more innovatory role in certain projects. For example, in March 1997, IASC's Steering Committee on Financial Instruments published proposals for dealing with financial assets and financial liabilities and those proposals went beyond any requirements or proposals that existed at a national level.
- 55. The development of innovative solutions will inevitably require greater resources, and may demand more significant changes to IASC's working methods, than was the case when IASC saw its main role as being to choose between existing national treatments. The Working Party believes that IASC should, in close partnership with national standard setters and other constituents, play an innovatory role in areas of increasing importance to IASC's constituents. Such areas may include:
  - (a) the growing use of new technology, such as the Internet and CD-ROMs, to deliver financial information in new ways. This may create a need for different or additional types of financial reporting standards;

- (b) emerging issues such as environmental reporting and accounting for human resources and intellectual capital; and
- (c) broader aspects of financial and other performance reporting outside the traditional financial statements, for example:
  - (i) financial reporting in a Management Discussion and Analysis ("MD&A"), Directors' Report, or similar document;
  - (ii) prospective financial information; and
  - (iii) non-financial measures of performance.

### WORKING WITH NATIONAL STANDARD SETTERS

- 56. The Working Party believes that IASC should, in developing International Accounting Standards, and in promoting their use, work closely with national standard setters to reach mutual agreement on what the highest quality result is. The aim is to ensure that national accounting standards and International Accounting Standards converge around high-quality solutions. The Working Party believes that IASC should work for convergence by:
  - (a) continuing to develop International Accounting Standards that build on the best features of existing and newly developed national standards. For topics where national standards do not yet exist, or are still evolving, IASC will need to work with national standard setters to develop high-quality requirements that lead to transparency and comparability;
  - (b) acting as a catalyst for, or initiator of, national developments in standard setting; and
  - (c) keeping existing International Accounting Standards under review in the light of the latest thinking at national and international levels. In some cases, this review may lead to the conclusion that a national standard provides greater transparency or comparability than an existing International Accounting Standard. In such cases, IASC will need to consider amending its existing Standard.

# CONVERGENCE OF STANDARDS FOR LISTED AND OTHER ECONOMICALLY SIGNIFICANT ENTERPRISES

- 57. Over the last few years, IASC has put emphasis on developing core standards that are suitable for cross-border capital raising and listing purposes. However, the Working Party believes that IASC should not restrict itself to developing standards for cross-border listings. Instead, IASC should develop standards for all listed and other economically significant enterprises, because:
  - (a) although certain countries may prefer for some time to require some enterprises (for example, those with cross-border listings) to use International Accounting Standards and to let other enterprises use a separate set of domestic standards, it will become increasingly difficult to justify the co-existence of two bodies of standards in the same country; and

- (b) rapid globalisation means that many enterprises have foreign providers of capital or other foreign users of their financial statements, even if the securities of those enterprises are not traded on foreign markets.
- 58. The Working Party believes that IASC should, in partnership with national standard setters, make every effort to accelerate convergence between national accounting standards and International Accounting Standards around solutions requiring listed and other economically significant enterprises in all countries to report high-quality, transparent and comparable information that will help participants in capital markets and others to make economic decisions. Although IASC and national standard setters have worked together successfully and narrowed the differences between accounting standards and procedures in different countries, the remaining differences cannot be eliminated overnight. In the Working Party's view:
  - (a) IASC's short-term aim should be for national accounting standards and International Accounting Standards to converge around high-quality solutions; and
  - (b) IASC's aim in the longer term should be global uniformity a single set of high-quality accounting standards for all listed and other economically significant business enterprises around the world. It is not possible to forecast how long this will take, as different countries are likely to converge with uniform global standards at different rates.

#### SMALL AND MEDIUM-SIZED ENTERPRISES (SMEs)

- 59. The Preface to International Accounting Standards and the Framework state that International Accounting Standards apply to the financial statements of all commercial, industrial and business reporting enterprises, in both the public and private sectors. However, IASC's top priority since 1995 has been to complete the package of "core" standards required by IOSCO for cross-border listings. Therefore, some people feel IASC is developing standards that are suitable only for large multinational enterprises and enterprises with cross-border listings.
- 60. Some have proposed that IASC should create exemptions for small and medium sized enterprises (SMEs), such as the exemptions found in the European Accounting Directives, or for non-listed enterprises. It is indeed possible that cost-benefit factors (acknowledged by the Framework as a constraint) may sometimes suggest one answer to an issue for large enterprises and a different answer for SMEs. Relevant factors for SMEs may include:
  - (a) a narrower range of users; and
  - (b) the fact that many SMEs are more constrained by tax and distribution rules as they are not part of a group and thus do not issue consolidated financial statements.
- 61. Currently, as the Board debates each proposed Standard, it considers whether different requirements (for example, simpler recognition and measurement rules or disclosure exemptions) are needed for particular classes of enterprises. In most cases, the Board has felt that its Standards should apply to all enterprises, but there are a few exceptions:

- (a) IAS 14, Segment Reporting, and IAS 33, Earnings Per Share, and the proposals in E62, Financial Instruments: Recognition and Measurement, apply only to enterprises with publicly traded equity or debt securities. In practice, this is also the case for IAS 34, Interim Financial Reporting, which applies if an enterprise is required or elects to publish an interim financial report under International Accounting Standards; and
- (b) IAS 15, Information Reflecting the Effects of Changing Prices, applied originally only to listed and other economically significant enterprises. (Since 1989, IAS 15 has been non-mandatory for all enterprises.)

# 62. The Working Party believes that:

- (a) regulators and standard setters in each country should decide, in the light of local circumstances:
  - (i) whether International Accounting Standards are appropriate for SMEs in that country;
  - (ii) how SMEs should be defined in that country; and
  - (iii) what accounting standards should be used by SMEs in that country; and
- (b) it is likely that many countries will choose to bring accounting standards for SMEs into line with International Accounting Standards. Therefore, IASC must be prepared to re-evaluate the entire package of International Accounting Standards and consider whether they are appropriate for SMEs.

# ECONOMIES IN TRANSITION AND DEVELOPING AND NEWLY INDUSTRIALISED COUNTRIES

- 63. Countries in transition to a market economy and developing and newly industrialised countries benefit from IASC's work because they do not need to devote their own scarce resources to developing financial reporting standards. However, some argue that IASC does not pay enough attention to the needs of users and preparers in these countries. Some of these countries are already represented on the Board and one current IASC project Agriculture is intended partly to help such countries.
- 64. As IASC's resources are limited, IASC could not devote enough time and effort to succeed with the IOSCO programme and at the same time carry out a comprehensive review to ensure that its Standards meet the needs of countries in transition and developing countries. Instead, IASC chose a more realistic strategy first concentrate on the IOSCO programme and then see whether more needs to be done for developing countries. Also, the growing acceptance of International Accounting Standards in industrialised countries is likely to increase IASC's credibility in developing countries.
- 65. For the reasons given earlier, the Working Party's belief is that, in time, all listed and other economically significant enterprises (including those in countries in transition and developing countries) should apply International Accounting Standards. Also, many developing countries have decided that International Accounting Standards are useful for SMEs.

66. In April 1998, the IASC Board approved a proposal for a project to investigate the accounting needs of economies in transition and developing and newly industrialised countries. The Working Party supports IASC's continuing investigations in this area.

#### PUBLIC SECTOR AND NOT-FOR-PROFIT ORGANISATIONS

- 67. IASC's Constitution does not limit IASC's objectives to financial reporting by business enterprises. However, to date, IASC has focused on financial reporting by business enterprises and has not dealt with:
  - (a) financial reporting in the public sector; and
  - (b) financial reporting by not-for-profit organisations, such as charities.
- 68. Currently, accounting and auditing standards for the public sector are being developed by the Public Sector Committee of IFAC, the International Federation of Accountants. The Public Sector Committee uses International Accounting Standards as a basis in developing its own accounting standards.
- 69. Some countries, notably Australia and New Zealand, are establishing common standards for both private and public sectors. The standard setters in those countries might find it more convenient if IASC were to take over responsibility for financial reporting in the public sector from IFAC. The Working Party acknowledges that this may be desirable in the longer term but notes the need to make the most effective use of IASC's limited resources. For this reason, the Working Party believes that IASC should continue to concentrate on business enterprises in the private sector for the time being and maintain a close dialogue with the Public Sector Committee.
- 70. The Working Party believes that IASC should not focus on financial reporting by not-for-profit organisations at this stage. However, it is likely to become important for IASC to address this topic at some point in the future.

**BLANK PAGE** 

# 4. IASC'S STRUCTURE AND DUE PROCESS TODAY

#### DESCRIPTION OF THE CURRENT STRUCTURE

- 71. The current version of IASC's Constitution was approved by the Members of IASC in 1992. Amendments to the Constitution must be discussed with the Council of IFAC and require a three-quarters majority of the IASC Board and approval by the Members of IASC as expressed by a simple majority of those voting.
- 72. IASC's Constitution sets out the powers of the Members and of the IASC Board. In addition, the IASC Board has established an Advisory Council, Steering Committees, a Standing Interpretations Committee and an Executive Committee. The roles of each of these bodies, and of the IASC staff and of IFAC, are discussed below.

#### **Member Bodies**

- 73. The Members of IASC are all those professional accountancy bodies that are Members of the International Federation of Accountants (IFAC). IASC currently has 143 Members in 103 countries. A meeting of the Members is held every two and a half years in conjunction with each General Assembly of IFAC.
- 74. Under IASC's Constitution, the Members have no direct role in the approval of International Accounting Standards or the appointment of the IASC Board.

#### IASC Board

- 75. Under the IASC Constitution, the Members of IASC have delegated the responsibility for all IASC's activities to the IASC Board. The Board has the power to:
  - (a) co-opt up to four organisations having an interest in financial reporting on to the Board:
  - (b) remove from membership of the Board any Board Member whose financial contribution is more than one year in arrears or which fails to be represented at two successive Board meetings;
  - (c) publish documents relating to international accounting issues for discussion and comment provided a majority of the Board votes in favour of publication;
  - (d) issue documents in the form of exposure drafts for comment (including amendments to existing standards) in the name of the International Accounting Standards Committee provided that at least two-thirds of the Board votes in favour of publication;
  - (e) issue International Accounting Standards provided that at least three-quarters of the Board votes in favour of publication;
  - (f) establish operating procedures so long as they are not inconsistent with the provisions of IASC's Constitution;
  - (g) enter into discussions, negotiations or associations with outside bodies and generally promote the world-wide improvement and harmonisation of accounting standards; and

- (h) seek and obtain funds from Members of IASC and non-members which are interested in supporting the objectives of IASC provided that such funding is organised in such a way that it does not impair the independence, or the appearance of independence, of IASC.
- 76. It follows from the powers granted to the Board that it is, at present, solely responsible for, among other things:
  - (a) determining the due process used to develop International Accounting Standards;
  - (b) choosing topics on its work programme and the priorities attached to those topics; and
  - (c) appointing Steering Committees, the Consultative Group, the Advisory Council and the Executive Committee.

# 77. The Board comprises:

- (a) thirteen country Members, appointed by the Council of IFAC after seeking the advice of the outgoing Board; and
- (b) up to four co-opted Members, appointed by the Board itself.
- 78. The thirteen country Members represent Members of IASC. To widen the membership of the Board, the Council of IFAC has appointed more than one country to share certain seats. Under the Constitution, the thirteen country Members are appointed by the Council of IFAC for up to five years, although recent appointments have been for two and a half years. Board Members may be re-appointed without limit.
- 79. Under Mutual Commitments agreed between IFAC and IASC, the Council of IFAC nominates to the IASC Board, after seeking the advice of the outgoing Board:
  - (a) at least nine of the most significant countries in terms of the status and development of the accountancy profession or that are of significant importance to international commerce and trade; and
  - (b) preferably, not less than three developing countries.

The Council of IFAC must satisfy itself that the nominees have standards and resources which would enable them to contribute to the work of the Board and are willing on the invitation of the Board to nominate persons to carry out assignments or to join Working Parties or groups constituted to undertake tasks allotted by the Board.

80. The co-opted Members are organisations with an interest in financial reporting. The terms of the co-opted Board Members are determined by the Board itself, with no upper limit.

81. For the two and a half year term ending 30 June 2000, the Board Members are:

<u>Country Members</u> (date of joining the Board in parentheses)

- Australia (1973)
- Canada (1973)
- France (1973)
- Germany (1973)
- India (1993) with Sri Lanka (1995)
- Japan (1973)
- Malaysia (1995)
- Mexico (1973-1988, re-appointed 1995)
- Netherlands (1973)
- Nordic Federation of Public Accountants (1988)
- South Africa (1978) with Zimbabwe (1995)
- United Kingdom (1973)
- United States of America (1973)

## Co-opted Members (one seat vacant)

- International Council of Investment Associations (1986)
- Federation of Swiss Industrial Holding Companies (1995)
- IAFEI International Association of Financial Executives Institutes (1996)
- 82. Each Board Member has one vote. Decisions are taken on a simple majority of the Board, except that a positive vote by two-thirds of all Board Members is required to approve an Exposure Draft and a positive vote by three-quarters of all Board Members is required to approve an International Accounting Standard, or, subject to the ratification of IASC's Members, to approve changes to the Constitution.
- 83. Each Board Member may nominate up to two representatives and a technical adviser to attend Board meetings. Neither the Board nor IASC can compel Board Members to nominate particular individuals, or individuals with particular characteristics. However, the Board encourages each Board Member to include in its delegation at least one person working in business and one person who is directly involved in the work of the national standard setting body.
- 84. The IASC Constitution states that Board Representatives shall not regard themselves as representing sectional interests but shall be guided by the need to act in the public interest.
- 85. The President of IFAC, or his designate, accompanied by not more than one technical adviser, is entitled to attend IASC Board meetings and may speak but not vote. Also, at the invitation of the Board, representatives of the following currently attend Board meetings as observers:
  - European Commission
  - United States Financial Accounting Standards Board (FASB)
  - International Organization of Securities Commissions (IOSCO)
  - Public Sector Committee of IFAC
  - China

- 86. The IASC Chairman is elected by the Board for a non-renewable term of two-and-a-half years. The Board has applied an informal policy that the Chairmanship should rotate between different regions. For continuity, the Board also normally appoints a Deputy Chairman who then takes up the Chairmanship for the next term.
- 87. The Board agreed in November 1997 to appoint two Vice-Chairmen for the next term, instead of a Deputy Chairman. The main reasons for this are to:
  - (a) avoid the need to identify a Chairman-elect at a time when, because of the current strategy review, the Chairman's role is under review; and
  - (b) spread the increasing burden of speaking commitments and other promotional and representational duties more widely.
- 88. The IASC Board used to meet three times a year for three or four days at a time. In 1996 the Board agreed to accelerate its work programme under an agreement with IOSCO. As a result, the Board now meets four times a year for five or six days at a time. This imposes heavy demands on the Board and it would be difficult for a Board made up of volunteers to continue meeting so often once the IOSCO work programme is completed.

## **Advisory Council**

- 89. In 1995, IASC established an Advisory Council to promote generally the acceptability of International Accounting Standards and enhance the credibility of IASC's work by, among other things:
  - (a) reviewing and commenting on the Board's strategy and plans so as to satisfy itself that the needs of IASC's constituencies are being met;
  - (b) preparing an annual report on the effectiveness of the Board in achieving its objectives and in carrying out its due process;
  - (c) promoting participation in, and acceptance of, the work of IASC by the accountancy profession, the business community, the users of financial statements and other interested parties;
  - (d) seeking and obtaining funding for IASC's work in a way that does not impair IASC's independence; and
  - (e) reviewing IASC's budget and financial statements.
- 90. Among other things, the Advisory Council aims to ensure that the independence and objectivity of the Board in making technical decisions on proposed International Accounting Standards are not impaired. The Advisory Council does not participate in, nor seek to influence, those decisions.
- 91. The members of the Advisory Council are outstanding individuals in senior positions from the accountancy profession, business and other users of financial statements from different backgrounds. The Advisory Council does not include representatives of national standard setters. The Chairman, Deputy-Chairman and Secretary-General of IASC are invited to, and expected to attend, all meetings of the Advisory Council but the Council is free to meet without them.

## **Steering Committees**

- 92. For its technical projects, the Board normally appoints a Steering Committee to:
  - (a) direct the preliminary research carried out by the staff;
  - (b) submit a Point Outline to the Board, in order to clarify the scope of the project;
  - (c) in the light of the Board's comments on the Point Outline, prepare and publish a Draft Statement of Principles;
  - (d) in the light of public comments on the Draft Statement of Principles, prepare and submit a Statement of Principles to the Board;
  - (e) based on the Statement of Principles approved by the Board, prepare and submit a draft Exposure Draft to the Board; and
  - (f) in the light of public comments on the Exposure Draft, prepare and submit a draft of the final International Accounting Standard to the Board.

For certain projects, the Board allows Steering Committees to omit one or more of stages (b), (c) and (d).

- 93. Steering Committees are chaired by a Board Representative and usually have around six to eight members. Most Steering Committee members are neither Board Representatives nor members of their national standard setter. The candidates for membership of a Steering Committee may be nominated by Members of IASC, other organisations that are represented on the Board or the Consultative Group and other organisations that are expert in the particular topic. In appointing Steering Committees, the Board seeks both a geographical balance and a mix of accountants in public practice, preparers and users. The Board also aims to ensure that the Steering Committee has sufficient specialist knowledge of the topic.
- 94. The Constitution states that members of Steering Committees shall not regard themselves as representing sectional interests but shall be guided by the need to act in the public interest.
- 95. The Board's current policy is to ask IOSCO to nominate an observer to attend each Steering Committee in which IOSCO has a special interest.

## **Standing Interpretations Committee**

96. In 1997, the IASC Board formed a Standing Interpretations Committee (SIC) to consider, on a timely basis, accounting issues that are likely to receive divergent or unacceptable treatment in the absence of authoritative guidance. Its review is within the context of existing International Accounting Standards and the IASC Framework. In developing interpretations, the SIC consults similar national committees which have been nominated for the purpose by Member Bodies.

- 97. The SIC deals with issues of reasonably widespread importance, and not issues of concern to only a small set of enterprises. The interpretations cover both:
  - (a) mature issues (the potential for unsatisfactory practice within the scope of existing International Accounting Standards); and
  - (b) emerging issues (new topics relating to an existing International Accounting Standard but not considered when the standard was developed).
- 98. The SIC has twelve voting members from various countries, including individuals from the accountancy profession, preparer groups and user groups. Some current members of the SIC are members of similar national bodies. IOSCO and the European Commission are non-voting observers. Because of the required time commitment, only two current members of the SIC are also Board Representatives. To ensure adequate liaison with the Board, two Board Representatives attend SIC meetings as non-voting Board Liaison Members.
- 99. The SIC publishes a draft interpretation, if no more than three of its voting members have voted against the draft. The public comment period for draft interpretations is two months. After considering comments received, the SIC amends the draft as it considers necessary. If no more than three of the SIC's voting members have voted against the final interpretation, the SIC asks the Board to approve the interpretation for formal publication; as for International Accounting Standards, this requires three-quarters of the Board to vote in favour.

## **Consultative Group**

100. In 1981, the IASC Board established an international Consultative Group that includes representatives of international organisations of preparers and users of financial statements, stock exchanges, regulators, trade unions, lawyers, valuers and other professions and academics. The group also includes representatives or observers from development agencies, standard setters and intergovernmental organisations.

## **Current Members of the Consultative Group**

**European Commission** 

Fédération Internationale des Bourses de Valeurs (FIBV)

Financial Accounting Standards Board (FASB)

International Actuarial Association (IAA)

International Association for Accounting Education and Research (IAAER)

International Association of Insurance Supervisors (IAIS)

**International Banking Associations** 

International Bar Association (IBA)

International Chamber of Commerce (ICC)

International Confederation of Free Trade Unions (ICFTU), and World

Confederation of Labour

International Finance Corporation (IFC)

International Valuation Standards Committee (IVSC)

The World Bank

Organisation for Economic Co-operation and Development (OECD)\*

United Nations Division on Transnational Corporations and Investment\*

\*Observers

- 101. The Consultative Group advises the Board on technical issues in specific IASC projects, on IASC's plans and priorities as set out in the work programme, on the likely acceptability of IASC's standards and on IASC's strategy. This group plays a helpful part in IASC's due process for the setting of International Accounting Standards and in gaining acceptance for the resulting standards.
- 102. The Consultative Group used to meet the full Board twice a year, for one day at a time. With the increasing numbers of participants, these meetings became less productive and the Consultative Group now meets selected Board Representatives, once a year for two days at a time.

#### **Executive Committee**

103. Management of administrative matters is in the hands of an Executive Committee appointed by the Board, headed by the Chairman of IASC and including the Vice-Chairmen and a number of other Board Representatives. The Executive Committee gives the Board a report on its discussions.

#### **IASC Staff**

- 104. The Board is supported by a small staff based in London, headed by a Secretary-General. The staff currently comprises:
  - (a) the Secretary-General (from the UK);
  - (b) a Technical Director (from Germany), from four to six other full-time (or almost full-time) technical staff and one part-time project manager. Current and recent past technical staff have been from Canada, France, Germany, Malaysia, New Zealand, South Africa, the UK and the USA. The technical staff are generally on secondments for one to two years, although some are employed on a longer-term basis; and
  - (c) a Commercial Director and nine other support staff.
- 105. Two recent IASC projects have been carried out jointly with national standard setters who have supplied project managers at no cost to IASC (other than direct travel costs).

## The role of IFAC

- 106. The accountancy profession plays a major part in supporting and funding the work of IASC. Professional accountancy bodies took the lead in establishing IASC because of the accountancy profession's important role at that time in the setting of national accounting standards in many countries. The support of the national and accountancy bodies and IFAC has contributed greatly to the acceptance of International Accounting Standards.
- 107. IASC's relationship with IFAC is set out in Mutual Commitments signed in 1982. In summary:
  - (a) all Members of IFAC are automatically also Members of IASC;

- (b) after seeking the advice of the outgoing Board, the Council of IFAC nominates the thirteen country Members of the IASC Board (see paragraphs 78 and 79 for further details);
- (c) the IASC Board is required to discuss with the Council of IFAC any proposed changes to the constitution of IASC. However, IFAC or IASC may amend their respective constitutions without the approval of the other body provided that such amendment is not in conflict with the substance of the Mutual Commitments (Amendments to IASC's Constitution require a three-quarters majority of the IASC Board and approval by the Members of IASC as expressed by a simple majority of those voting);
- (d) the President of IFAC or his designate is entitled to attend, and speak at, IASC Board meetings and the IASC Chairman may attend, and speak at, meetings of the Council of IFAC;
- (e) IFAC recognises IASC as the sole body having responsibility and authority to issue, in its own name, pronouncements on international accounting standards with full authority in so doing to negotiate and associate with outside bodies and to promote the worldwide acceptance and observance of those standards. IFAC shall not appoint or support any other body for this purpose and shall not itself formulate or consider or publish any other such standards but shall support the standards promulgated by IASC and shall require its Members to support the work of IASC by publishing in their respective countries every International Accounting Standard approved for issue by the Board of IASC and by using their best endeavours:
  - (i) to ensure that published financial statements comply with International Accounting Standards in all material respects and disclose the fact of such compliance;
  - (ii) to persuade governments and standard-setting bodies that published financial statements should comply with International Accounting Standards in all material respects;
  - (iii) to persuade authorities controlling securities markets and the industrial and business community that published financial statements should comply with International Accounting Standards in all material respects and disclose the fact of such compliance;
  - (iv) to ensure that the auditors satisfy themselves that the financial statements comply with International Accounting Standards in all material respects; and
  - (v) to foster acceptance and observance of International Accounting Standards internationally;
- (f) IFAC contributes one ninth of the budgeted net expenditure that the Board determines should be borne by Board members. Also, IFAC currently reimburses the cost of one Board seat. This subsidy is shared equally by India, Sri Lanka and Zimbabwe; and

(g) changes to the Mutual Commitments require both the approval of the Council of IFAC and three-quarters of the total of the votes of the Board of IASC.

#### **DUE PROCESS**

- 108. In the absence of specific provisions in the Constitution, it is the responsibility of the Board to establish the due process for developing standards. IASC's due process is designed to ensure that International Accounting Standards are of high quality and acceptable to the users and preparers of financial statements. The current due process is as follows:
  - (a) the Board sets up a Steering Committee;
  - (b) the Steering Committee identifies and reviews all the accounting issues associated with the topic. The Steering Committee considers the application of IASC's Framework for the Preparation and Presentation of Financial Statements to those accounting issues. The Steering Committee also studies national and regional accounting requirements and practice, including the different accounting treatments that may be appropriate in different circumstances. Having considered the issues involved, the Steering Committee may submit a Point Outline to the Board, in order to clarify the scope of the project;
  - (c) after receiving comments from the Board on the Point Outline, if any, the Steering Committee normally prepares and publishes a Draft Statement of Principles or other discussion document. The purpose of this Statement is to set out the underlying accounting principles that will form the basis for the preparation of an Exposure Draft. It also describes the alternative solutions considered and the reasons for recommending their acceptance or rejection. Board approval is not required for the Draft Statement of Principles. Comments are invited from all interested parties during the exposure period, usually around three months. The Board may instruct the Steering Committee to prepare an Exposure Draft without first publishing a Draft Statement of Principles;
  - (d) the Steering Committee reviews the comments on the Draft Statement of Principles and normally agrees a final Statement of Principles, which is submitted to the Board for approval and used as the basis for preparing an Exposure Draft of a proposed International Accounting Standard. The final Statement of Principles is available to the public on request, but is not formally published;
  - (e) the Steering Committee prepares a draft Exposure Draft for approval by the Board. After revision, and with the approval of at least two-thirds of the Board, the Exposure Draft is published. Comments are invited from all interested parties during the exposure period, a minimum of one month and usually at least three months; and
  - (f) the Steering Committee reviews the comments and prepares a draft International Accounting Standard for review by the Board. After revision, and with the approval of at least three-quarters of the Board, the standard is published.

- 109. Neither Steering Committee meetings nor Board meetings are open to the public. However, the Board agreed in November 1998 that Board meetings should be open to the public from 1999.
- 110. During this process, the Board may decide that the needs of the subject under consideration warrant additional consultation or would be better served by issuing a Discussion Paper for comment. Under the IASC Constitution, the Board has the power to publish documents relating to international accounting issues for discussion and comment provided that a majority of the Board votes in favour. In some cases, Steering Committees or the staff have issued Discussion Papers or Issues Papers without formal Board approval.

# 5. IMPROVING IASC'S STRUCTURE AND DUE PROCESS

111. This section considers how IASC's structure and due process may need to be adapted to respond to the new challenges identified earlier in this Discussion Paper.

#### **EVALUATION OF THE CURRENT STRUCTURE**

- 112. The current structure of IASC has significant strengths:
  - (a) IASC has produced high-quality standards that command international support, without unnecessary delay, generally by using existing national standards as a starting point;
  - (b) the geographical spread of Board membership, and the requirement that a final standard must achieve a positive vote from three-quarters of the Board as currently constituted mean that IASC must persuade a reasonably broad constituency that its proposals are appropriate an important consideration for an organisation that cannot compel countries or individual enterprises to adopt its standards. At the same time, the required majority is not so high that progress is blocked;
  - (c) most Board delegations are currently made up of three individuals (two Board Representatives and one Technical Adviser). This permits a functional mix (preparers, auditors, standard setters, financial analysts, academics and others) from most countries on the Board and gives the Board the broad perspective that comes from a diversity of backgrounds;
  - (d) the part-time status of Board Representatives and technical advisers enables them to stay in touch with their constituents and to retain up-to-date experience of accounting practice in their countries;
  - (e) continuity of Board membership (both delegations and their individual representatives) speeds progress, promotes consistency and builds an atmosphere of collegiality and trust which is very important;
  - (f) the involvement of a wide range of people in IASC's process, through both the Board itself and Steering Committees, plays an important promotional and educational role for IASC; and
  - (g) IASC functions at remarkably low direct cost.
- 113. IASC has achieved a great deal with the current structure. It has developed high-quality and credible standards. Its standards are widely accepted by the international capital markets. A growing number of countries are either adopting International Accounting Standards as their own standards (in some cases, with relatively minor modifications) or drastically reducing provisions in their own standards that conflict with International Accounting Standards.
- 114. Despite the strengths of IASC's current structure, the changes in IASC's environment mean that structural changes are needed so that IASC can anticipate the new challenges facing it and meet those challenges effectively. The Working Party has identified the following key issues that must be addressed:

- (a) partnership with national standard setters IASC should enter into a partnership with national standard setters so that IASC can work together with them to accelerate convergence between national standards and International Accounting Standards around solutions requiring high-quality, transparent and comparable information that will help participants in capital markets and others to make economic decisions;
- (b) wider participation in the IASC Board a wider group of countries and organisations should take part in the IASC Board, without diluting the quality of the Board's work.
- (c) appointment the process for appointments to the IASC Board and key IASC committees should be the responsibility of a variety of constituencies, while ensuring that those appointed are competent, independent and objective.

The Working Party also believes that changes are needed to IASC's due process so that it can cope effectively with the new challenges before it. The Working Party's proposals for improving IASC's due process are set out in paragraphs 201-214.

## **Partnership with National Standard Setters**

- 115. National standard setters have been a major factor in IASC's success. They have:
  - (a) helped to generate the diversity of ideas that is needed to develop the innovative solutions required in today's rapidly changing environment;
  - (b) played an important role in identifying emerging issues or problems with existing standards on a timely basis;
  - (c) facilitated the flow of information between their domestic constituents and IASC; and
  - (d) contributed generous human and technical resources to IASC.
- 116. The Working Party considers that the continued existence of strong and autonomous national standard setters will be an important ingredient in IASC's future success. The Working Party also recognises that many countries have regulations that require their national standard setters to maintain their autonomy. In addition, national standard setters will need to deal with issues that are important locally but not important enough to require priority treatment internationally.
- 117. An increasingly important challenge for IASC will be to work with national standard setters to bring about convergence between national accounting standards and International Accounting Standards around solutions requiring listed enterprises (i.e. enterprises with publicly traded equity or debt securities) and other economically significant enterprises to report high-quality, transparent and comparable information that will help participants in capital markets and others to make economic decisions. The standards of many countries are already converging with International Accounting Standards. However, trends such as globalisation and the increasing pace of business and financial change have made this task more urgent. The Working Party believes that IASC and national standard setters need to find new ways of working together to minimise unnecessary delays in reaching consensus and implementing the results of that consensus.

- 118. IASC already works informally with national standard setters. For example, many Board Members include in their delegations at least one person who is involved in the work of the national standard setter. However, a closer and more formal partnership between IASC and national standard setters would have important benefits:
  - (a) national standard setters and IASC would find it easier to co-ordinate their work programmes with each other. This would reduce the risk that different standard setters will address the same issue at different times and reach different solutions to the same issue;
  - (b) national standard setters would feel that they have some "ownership" of the standards that emerge from the IASC process. This would increase the acceptability of IASC's standards to national standard setters and their constituents;
  - national standard setters would be more likely than before to publish Exposure Drafts that are identical, or very similar, to IASC Exposure Drafts, to complete consultation with their own constituents before IASC needs to finalise its own standard and to play a full part in the IASC process. In some cases they may consider that they have compelling national reasons for maintaining or developing differences from International Accounting Standards. However, a close partnership between IASC and the national standard setters should mean that such differences become increasingly rare; and
  - (d) assistance from national standard setters in preparing Exposure drafts and Standards would greatly augment IASC's resources and result in more efficient use of the limited resources available around the world for standard setting.
- 119. Given the Working Party's view that national standard setters should have a more formal role in IASC, it seems natural to review the role of Steering Committees.
- 120. Steering Committees are relatively small groups and, despite the best efforts of the Board to secure a balanced membership, their composition is somewhat haphazard. This means that a relatively small change in the membership of a Steering Committee may have a large impact on its recommendations. Also, Steering Committees are made up of volunteers. Those volunteers may lack the time and the support needed to consider complex technical issues in sufficient depth and to understand how decisions in their projects interact with other current projects. These difficulties could be resolved by involving national standard setters more directly in preparing Exposure Drafts and Standards, a task that is currently done by Steering Committees.
- 121. The Board itself is too large to take over the work currently done by Steering Committees in preparing the detailed content of Standards. The Board comprises sixteen members, each normally represented by three individuals (two Board Representatives and one Technical Adviser). Together with observers, occasional guests and the staff, there are now usually over 70 people at the Board table. This suggests that the detailed preparation of standards should continue to be the responsibility of a smaller group than the Board. The Working Party believes that national standard setters should be closely involved in this smaller group.

## Wider Participation in the IASC Board

- 122. The Working Party believes that IASC should widen the composition of the Board, for the following reasons:
  - (a) IASC has achieved much greater importance and higher profile in recent years. Also, its old role as a harmoniser is evolving into a new role as an innovator. These factors mean that many more countries and organisations want a voice at the Board table; and
  - (b) developing countries and countries in transition have limited representation at the Board table (for example, China is only an observer and the countries of the former Soviet Union, Central and Eastern Europe and South America have no representation on the Board). Therefore, there is a risk that the Board will not take sufficient account of their needs and views.

## **Appointment**

- 123. The Working Party believes that IASC needs to amend the procedure for appointing organisations and individuals to its decision-making bodies, for the following reasons:
  - (a) appointments to the country seats on the IASC Board are in the hands of the accounting profession (through IFAC). Therefore, preparers, users and others feel that they are under-represented and that the Board is dominated by the accountancy profession;
  - (b) Board Members choose the individuals who will serve as their Board Representatives and Technical Advisors. Thus, there is no guarantee of a proper balance in the Board as a whole of individuals from public accounting, the business community and other users of financial statements as well as individuals who are directly involved in the work of the national standard setters; and
  - (c) a key responsibility of the Advisory Council is to monitor the Board. However, the Advisory Council may appear to lack objectivity because it is appointed by, and reports to, the Board itself.

### A POSSIBLE FUTURE STRUCTURE FOR IASC

- 124. The Working Party's proposals are set out in paragraphs 125-200 below.
- 125. The Working Party's proposals address the key issues identified in paragraph 114 by the following changes:
  - (a) a partnership with national standard setters:
    - (i) Steering Committees would be replaced by a Standards Development Committee, on which national standard setters would play a major role in developing International Accounting Standards (see paragraphs 126-136 for its membership and paragraphs 154-159 for its responsibilities). The Standards Development Committee would also be responsible for approving the publication of final SIC Interpretations prepared by the Standing Interpretations Committee; and

- (ii) the Standards Development Committee would be supported by a Standards Development Advisory Committee, which would act as a channel of communication with those national standard setters who are unable to participate directly in the Standards Development Committee because of its limited size (see paragraphs160-163);
- (b) wider participation in the IASC Board the Board would have a wider membership than at present (see paragraphs 137-146 for its membership and paragraphs 164-178 for its responsibilities). The Board would still be responsible for the final approval of International Accounting Standards and Exposure Drafts; and
- (c) appointment the Advisory Council would be replaced by Trustees (see paragraphs 147-153 for their membership and paragraphs 179-183 for their responsibilities). Among other things, the Trustees would appoint members of the Standards Development Committee, the Board and the Standing Interpretations Committee. The Trustees would also have responsibility for monitoring IASC's effectiveness and for finance.

Paragraphs 184-200 deal with other aspects of the Working Party's proposals for the structure of IASC. Paragraphs 201-214 cover the Working Party's recommendations for improving IASC's due process.

## **Standards Development Committee - Membership**

- 126. The Working Party proposes that Steering Committees should be replaced by a single Standards Development Committee to work on all IASC projects. The Standards Development Committee would prepare drafts of Exposure Drafts and International Accounting Standards for submission to the Board.
- 127. To allow for effective and efficient decision-making and ensure sufficient diversity of backgrounds, the Standards Development Committee should have 11 members, appointed by the Trustees:
  - (a) a full-time chairman;
  - (b) six to eight individuals, each nominated by a national standard setter with the technical, human and financial resources to make a significant contribution to IASC's work. The individuals should be voting members of the national standard setter that nominated them; and
  - (c) two to four members selected from other groups, such as preparers, users, accountants in public practice and academics.
- 128. Criteria for membership of the Standards Development Committee should include the following:
  - (a) prospective members should be people of proven technical competence in standard setting, integrity and objectivity. They should not regard themselves as representing sectional interests but should be guided by the need to act in the public interest. Therefore, they would reach their own conclusions on technical issues and not merely vote on behalf of their national standard setter or other constituency;

- (b) prospective members should accept a commitment to use IASC's Framework for the Preparation and Presentation of Financial Statements as the basis for developing Standards that require high-quality, transparent and comparable information which will help a wide range of users (including participants in the capital markets) to make economic decisions. Revisions to the Framework should go through the same due process as a new Standard;
- (c) prospective members should show, for example, by previous participation in IASC projects and by submitting comment letters on IASC proposals, that they can make an active contribution to IASC's work and that they will be able to attend meetings regularly.
- (d) if suitably qualified candidates are available, at least seven should be from more developed countries, for example, countries that are members of the OECD or other countries with large economies or large capital markets (see appendix 5 for current data) and at least two should be from developing countries or countries in transition to a market economy;
- (e) special consideration should be given to individuals from countries that:
  - (i) make, or are likely to make, extensive use of International Accounting Standards; or
  - (ii) have a high proportion of enterprises with foreign operations or that are involved in foreign trade; and
- (f) there should be a reasonable geographical spread of members.
- 129. The Working Party believes that all members of the Standards Development Committee should devote at least half of their time to standard setting, including any time spent at their national standard setter. This would:
  - (a) provide greater independence from employment and other allegiances that might affect their voting; and
  - (b) increase their standard-setting expertise and the time they can give to technical issues.
- 130. The Working Party believes that the Standards Development Committee will need a full-time Chairman. The Working Party believes that at least six other members of the Standards Development Committee should have a full time involvement in standard setting, including time spent at their national standard setter.
- 131. Most national standard setters have mainly part-time members and would probably need to appoint a full-time member specifically to serve on the Standards Development Committee. The work of the Standards Development Committee would be of direct benefit to national standard setters and other constituencies. Therefore, the Working Party believes that national standard setters and other constituencies should normally pay:
  - (a) the salaries of their own members on the Standards Development Committee and related costs, including insurance;
  - (b) the costs of their travel between IASC and the national standards body; and

- (c) the costs of technical and other staff assigned by the national standard setter to assist the member. However, IASC should pay for the costs of IASC technical and other staff.
- 132. Some countries may find it difficult to identify, and finance, a suitable individual to serve on the Standards Development Committee. The Trustees may need to seek a special arrangement with professional accountancy bodies in these countries to identify, and finance, a suitable individual. However, the Working Party recognises that it may be difficult to achieve such an arrangement.
- 133. Certain countries or other constituencies may not be able to finance a member of the Standards Development Committee. The Trustees should have the power to direct IASC to pay for the costs of well-qualified individuals who would enhance the diversity of backgrounds in the Standards Development Committee.
- 134. The Standards Development Committee would probably need to meet every one to two months. The use of video conferencing and e-mail may reduce the need for all members to travel to every meeting. However, most members should travel to most meetings, as face-to-face meetings are essential to ensure full communication and to build an atmosphere of collegiality and trust. To minimise logistical problems for members based outside Europe or North America, meetings may need to rotate.
- 135. Standards Development Committee members should be appointed for a fixed term of five years. However, the expiry of the first terms should be staggered: two members should retire after each of three, four, six and seven years and three members should retire after five years. This will provide continuity when the Standards Development Committee is first set up. The terms of individual members should be renewable once at the discretion of the Trustees and another individual from the same country or organisation would be eligible for subsequent appointment.
- 136. As stated in paragraph 204, the Working Party believes that the technical discussions of the Standards Development Committee should be open to the public. The Standards Development Committee would be free to invite members of the public to speak at its meetings. The Working Party leans towards the view that the Standards Development Committee should not have formal observers with a right to speak, but recognises that the question of regulatory representation may need further thought.

## **Board - Membership**

- 137. In some national standard setters, decisions are taken by a relatively small group of, perhaps between seven and thirteen, experts. In an international standard setter, it may be desirable to give more constituencies (both geographical and functional) representation in the body that has ultimate authority for approving Standards. The Working Party believes that the Board should have a wider membership than at present:
  - (a) 20 country seats for professional accountancy bodies (rather than thirteen as at present); and
  - (b) five seats for other organisations with an interest in financial reporting (at present, four seats are reserved for such organisations and three are currently occupied). These should generally be international organisations, although exceptions should be permitted in limited cases.

- 138. Criteria for membership of the Board should include the following:
  - (a) at least fourteen of the country Members should be more developed countries, for example, countries that are members of the OECD or other countries with large economies or large capital markets (see appendix 5 for current data) and at least four should be developing countries or countries in transition.
  - (b) special consideration should be given to countries that:
    - (i) make, or are likely to make, extensive use of International Accounting Standards; or
    - (ii) have a high proportion of enterprises with foreign operations or that are involved in foreign trade;
  - (c) there should be a reasonable geographical spread of Board Members;
  - (d) there should be a planned rotation in Board Membership that provides a reasonable balance between continuity and turnover; and
  - (e) Board Members should show that they can make an active contribution to the Board, for example, by previous participation in IASC projects and by submitting comment letters on IASC proposals.
- 139. The IASC Constitution states that Board Representatives shall not regard themselves as representing sectional interests but shall be guided by the need to act in the public interest. To emphasise this point, the individuals appointed by Board Members should be described as Board Delegates and not Board Representatives.
- 140. To keep discussions effective, each Board delegation should be represented by two Board Delegates, rather than the three (two Board Representatives and one Technical Adviser) individuals that form most present Board delegations.
- 141. At present, each delegation has one vote. This means that Board delegations are sometimes forced to abstain where the members of the delegation are unable to agree among themselves. Given the current requirement for a positive vote by three quarters of the Board, an abstention is effectively the same as a vote against a Standard. This might suggest that each member of the delegation should be given an individual vote, to reduce the risk of deadlock. However, discussions among, say, 50 voting individuals would be much more cumbersome than discussions among 50 individuals representing 25 voting delegations. The Working Party recommends that IASC should retain the current practice that each delegation has one vote.
- 142. Board Members should retain the power to appoint their Board Delegates, in consultation with the Trustees, who would have the power to veto the appointment of any individual Board Delegate. The Working Party believes that the Trustees should exercise this power extremely rarely and proposes that such a veto should require the agreement of nine of the twelve Trustees. The Trustees should develop and apply written criteria to ensure that:
  - (a) Board Delegates are people of technical competence, integrity and objectivity who will not regard themselves as representing sectional interests, but will be guided by the need to act in the public interest;

- (b) Board Delegates will accept a commitment to use IASC's Framework for the Preparation and Presentation of Financial Statements as the basis for developing Standards that require high-quality, transparent and comparable information which will help a wide range of users (including participants in the capital markets) to make economic decisions. Revisions to the Framework should go through the same due process as a new Standard;
- (c) the Board has sufficient diversity of membership, including accountants in public practice, accountants in industry, commerce and finance, people involved in national standard setting (particularly for those countries whose standard setter is not represented on the Standards Development Committee), users, academics and others;
- (d) Board Delegates can make an active contribution to IASC's work and will attend meetings regularly; and
- (e) there is a planned rotation of Board Delegates that provides a reasonable balance between continuity and turnover.
- 143. All Members of the Standards Development Committee should attend Board meetings and have a right to speak at those meetings. However, to avoid conflicts of interest:
  - (a) individuals who are members of the Standards Development Committee should not be eligible to serve as Board Delegates or to act as voting alternates to Board Delegates, although another individual from the same country should be eligible; and
  - (b) similarly, where an individual is a member of the Standards Development Committee and also a member of a national standard setter, another individual from the same standard setter should not be eligible to serve as a Board Delegate or to act as a voting alternate to a Board Delegate.

Where an individual is a member of the Standards Development Committee and also a member of an organisation other than a national standard setter, another individual from the same organisation should be eligible to serve as a Board Delegate.

- 144. Members of the Board should be appointed for a fixed term of two and a half years, renewable without limit at the discretion of the Trustees. Given the proposed criterion of a reasonable balance between continuity and turnover in Board Membership, there is no need to stagger the expiry of the terms.
- 145. The Board would be chaired by a non-executive, part-time Chairman, devoting 50% of his or her time to IASC. IASC would provide secretarial support and a half-time salary. The Chairman of the Board would be the main spokesperson for IASC on matters of broad policy and on the acceptability of International Accounting Standards to IASC's constituents. The position would be similar in some respects to the non-executive chairman of a business enterprise.
- 146. The Working Party proposes that the Trustees should have the power to appoint Observers with a right to speak, but not vote, at Board meetings. Observers would be regulators and other organisations who can make a contribution to the Board's debate but who are unable to take up a voting seat. There should be no formal limit on the number of observers, but the number should be kept relatively low to keep Board

meetings to a manageable size. The Working Party envisages that the Observers would include:

- (a) IOSCO;
- (b) the European Commission; and
- (c) IFAC and IFAC's Public Sector Committee and International Auditing Practices Committee.

## **Trustees - Membership**

- 147. At present, the accountancy profession (through IFAC) controls the appointments to the IASC Board. Because of the increasing importance of IASC's work, the Working Party believes that other constituents should have some say in appointments, although the accountancy profession (IFAC and the existing Member Bodies of IASC and IFAC) should retain a significant ongoing role.
- 148. The Working Party proposes that the Advisory Council should be replaced by Trustees with powers and responsibilities assigned by a revised IASC Constitution, rather than delegated to them by the Board. These would include the responsibility for appointments.
- 149. The Working Party proposes that there should be 12 Trustees, who would be unpaid except for the Chairman of the Trustees (to be paid on a part-time basis).
  - (a) Six "constituency" Trustees should be appointed by various constituents, in consultation with the existing Trustees:
    - three by the International Federation of Accountants (IFAC);
    - three by other international organisations, such as those currently represented on the Consultative Group.
  - (b) Six Trustees should represent the world "at large". The first six "at large" Trustees should be appointed by a nominating committee comprising the current Advisory Council, recent past chairmen of IASC and recent past presidents of IFAC (but excluding any individuals who will be candidates to serve as Trustees). For subsequent terms, the six "at large" Trustees should be appointed by the twelve Trustees.
- 150. To guide constituents in nominating the "constituency" Trustees and to guide themselves in selecting the "at large" Trustees, the Trustees should develop written criteria, including the following:
  - (a) Trustees should be people of integrity and objectivity who will be guided by the need to act in the public interest. Neither "constituency" Trustees nor "at large" Trustees should regard themselves as representing sectional interests;
  - (b) Trustees should accept a commitment to maintaining IASC as an organisation that develops Standards that require high-quality, transparent and comparable information which will help a wide range of users (including participants in the capital markets) to make economic decisions;

- (c) most of the twelve Trustees should be from more developed countries, for example, members of the OECD or other countries with large economies or large capital markets (see appendix 5 for current data), but it is desirable that some should be from developing countries or countries in transition to market economies:
- (d) special consideration should be given to Trustees from countries that:
  - (i) make, or are likely to make, extensive use of International Accounting Standards; or
  - (ii) have a high proportion of enterprises with foreign operations or that are involved in foreign trade;
- (e) there should be a reasonable geographical spread of Trustees;
- (f) collectively, the Trustees should have sufficient diversity of membership, including people with current or recent experience in senior positions as users, preparers, accountants in public practice, academics, regulators and others; and
- (g) Trustees should be able to make an active contribution to IASC's work and should attend meetings regularly.
- 151. The Trustees should appoint Chairman of the Trustees. The first Chairman should be designated by the nominating committee discussed above.
- 152. To avoid conflicts of interest, the Chairman of the Trustees should not be the same individual as the Chairman of the Board. Similarly, Trustees should not serve as members of the Standards Development Committee, as Board Delegates or as members of the Standing Interpretations Committee.
- 153. The Trustees would be appointed for a fixed term of five years (renewable once). The terms of the Trustees should be staggered so that they do not all retire at the same time. Thus, three of the first "constituency" Trustees and three of the first "at large" Trustees should retire after two and a half years.

#### **Standards Development Committee - Responsibilities**

- 154. The most important function of the Standards Development Committee will be to prepare Exposure Drafts and International Accounting Standards and submit them to the Board for approval. This should require approval by a "super-majority" of the Standards Development Committee (seven votes out of 11). The Standards Development Committee should also have responsibility for approval of SIC final Interpretations (by the same majority seven votes out of 11), as explained in paragraph 185. All other decisions should require a simple majority of those present.
- 155. The Standards Development Committee would have the power to add projects to, or remove them from, its work programme. Among other things, this will make it easier for IASC to co-ordinate work plans with national standard setters, which is an important factor in the partnership that IASC needs to forge with national standard setters.
- 156. The Working Party believes that the Standards Development Committee should have the authority, on a simple majority vote, to issue Draft Statements of Principles, as

- well as Discussion Papers and similar documents without Board approval, although it should discuss key issues with the Board in advance. This is similar to the authority currently given to Steering Committees.
- 157. The close involvement of national standard setters would mean that many IASC projects could be carried out as joint projects with national standard setters. In the past, national standard setters have borne their own staff costs for joint projects, including the cost of time spent on developing IASC documents for Steering Committee and Board meetings and for publication. The Working Party believes that IASC should continue to seek the agreement of national standard setters to co-operate on this basis.
- 158. Explanation and promotion is a major time commitment for national standard setters. To limit the demands on the time of the Chairman of the Standards Development Committee, the Chairman of the Board should carry out many of the representative duties as ambassador for IASC. The Chairman and other members of the Standards Development Committee should also be expected to help in explaining and promoting the technical aspects of IASC's work. The Working Party hopes that national standard setters would also play a part in this, by publishing and explaining IASC proposals as a vital component of their own domestic standard setting programme.
- 159. The Standards Development Committee should work in whatever way it considers most effective and cost-efficient. It may wish to set up sub-groups, task forces or similar bodies for individual projects or to outsource detailed research or other work to national standard setters. The Standards Development Committee should seek agreements with national standard setters to draw on their resources, as well as relying on IASC's own staff. It may also need to set up sub-committees or advisory groups to advise it on such issues as:
  - (a) accounting in specialist industries, such as banking, insurance, agriculture and the extractive industries;
  - (b) the reporting needs of unlisted enterprises or other small and medium-sized enterprises (SMEs) and not-for-profit organisations;
  - (c) the needs of developing countries and countries in transition;
  - (d) real-time reporting and the delivery of financial information through new technology, such as the Internet and CD-ROMs;
  - (e) emerging issues such as environmental reporting and accounting for human resources and intellectual capital; and
  - (f) broader aspects of financial and other performance reporting outside the traditional financial statements (for example, non-financial measures of performance and historic and prospective financial information in a Management Discussion and Analysis, Directors' Report, or similar document).

# Involving Other National Standard Setters - a Standards Development Advisory Committee

- 160. The need to limit the size of the Standards Development Committee means that it cannot include representatives from all standard setters in all countries. The inclusion of standard setters of the countries with the largest capital markets and a reasonable cross-section of other standard setters is likely to be a strong incentive for other standard setters to make use of IASC's work. Nevertheless, the Working Party believes that those other standard setters will need a formal channel of communication with the Standards Development Committee.
- 161. To provide this formal channel of communication, the Working Party recommends that there should be a Standards Development Advisory Committee. This should meet at least annually under the chairmanship of the Chairman of the Standards Development Committee. The purpose of these meetings should be to advise the Standards Development Committee whether its proposals are likely to be appropriate and operational in the domestic environment of the countries concerned. Membership of the Standards Development Advisory Committee should be at the invitation of the Trustees and should be open in principle to standard setters of any country not represented on the Standards Development Committee.
- 162. Among other things, the Standards Development Advisory Committee would provide a useful means for the Standards Development Committee to communicate with standard setters in countries in transition and developing and newly industrialised countries.
- 163. The Working Party believes that the meetings of the Standards Development Advisory Committee would replace the meetings of world standard setters that currently take place from time to time with IASC involvement.

## **Board - Responsibilities**

- 164. The Working Party paid close attention to the question of the final authority to issue standards. In its discussions, the Working Party identified three crucial considerations:
  - (a) the need to convince users, preparers and IASC's other constituents that IASC's standards will meet their needs;
  - (b) the need to attract suitably qualified individuals to serve on the Standards Development Committee and the Board; and
  - (c) the need for the Standards Development Committee and the Board to work together closely and effectively for the public interest.
- 165. IASC cannot force anyone to use its Standards and so must rely on persuasion. It can persuade its constituents to use its Standards only if the Standards are of high quality and meet their needs. Also, IASC's constituents are more likely to use its Standards if they have a stake in, and play a meaningful part in, their development and participation by IASC's constituents is likely to improve the quality of the standards.

- 166. One way to persuade IASC's constituents to accept its due process and its standards would be to set up an autonomous body of independent full-time and highly skilled experts, with a relatively small number of members for the sake of efficiency (an independent expert model). Another route would be to create a more broadly-based group from a larger number of countries and backgrounds (a constituency model).
- 167. The Working Party believes that neither of these extremes would secure sufficient worldwide support from IASC's constituents, even with the improvements that the Working Party is proposing to IASC's due process. Instead, the Working Party's proposal combines elements of both models: a group of independent experts (the Standards Development Committee) and a broader group (the Board), coupled with a high-level of due process to ensure a wide range of input.
- 168. In developing its proposals, the Working Party was conscious of the need to attract talented and well-qualified individuals to serve on both the Standards Development Committee and the Board. It is unlikely that such individuals will make themselves available for a body that does not have genuine decision-making power.
- 169. It is clear that any structure that involves two bodies with genuine decision-making power can work only if the two bodies demonstrate a clear willingness to work together closely and effectively for the public interest. The Working Party believes that a constructive attitude of close and effective co-operation is a striking feature of the current Board and of IASC's Committees. Consequently, the Working Party is confident that the Standards Development Committee and the Board will achieve the close and effective relationship that will be needed. The Working Party also noted that constructive co-operation will not emerge if either the Standards Development Committee or the Board is unable to attract suitably qualified individuals.
- 170. With these three considerations in mind, the Working Party considered a range of different ways of specifying the respective powers of the Standards Development Committee and the Board. Among these were:
  - (a) positive approval required by a majority or super-majority of the Board for all International Accounting Standards and Exposure Drafts;
  - (b) the Board (or perhaps a specified majority or a specified minority of the Board) has the power to reject proposed International Accounting Standards and Exposure Drafts;
  - (c) the Board (or perhaps a specified majority or a specified minority of the Board) has the power to return proposed International Accounting Standards and Exposure Drafts to the Standards Development Committee for re-consideration, but not to reject them indefinitely; and
  - (d) the Standards Development Committee must consult the Board, but the Board has no power to delay or reject International Accounting Standards or Exposure Drafts.
- 171. Individual members of the Working Party have preferences for different points in this range. Some members of the Working Party prefer option (a) above. Other members of the Working Party prefer option (d) above. However, the Working Party believes that the precise voting arrangements are less important than the need for the Standards Development Committee and the Board to work together constructively. As

- explained above, the Working Party is confident that the Standards Development Committee and the Board will achieve the close and effective relationship that will be needed and will maintain a close dialogue throughout each project.
- 172. The Working Party proposes that publication of a Standard or Exposure Draft should require approval by 60% of the Board (15 votes out of 25). At present, an Exposure Draft requires a positive vote by two thirds of the Board; a final Standard requires a positive vote by three quarters of the Board. The Working Party further concluded that the Chairman of the Board should be required to ensure that the Board considers and votes on all proposed Exposure Drafts and Standards submitted by the Standards Development Committee within three months of receipt or, if later, at its next meeting.
- 173. If the Board rejects a proposed Exposure Draft or Standard, the Board should send the document back to the Standards Development Committee for further consideration, giving public reasons for its rejection. After considering the reasons given by the Board, the Standards Development Committee may decide to:
  - (a) prepare a revised proposal and submit it to the Board for approval in the normal way; or
  - (b) resubmit its original proposal to the Board:
    - (i) if nine or more members of the Standards Development Committee have voted to resubmit the same proposal, Board approval should require a simple majority (13 votes out of 25); and
    - (ii) if seven or eight members of the Standards Development Committee have voted to resubmit the proposal, the proposal should be treated in the same way as a new proposal. In other words, Board approval should require the normal 60% majority (i.e. 15 votes out of 25).
- 174. The Working Party believes that the Board should not have the power to amend proposed Exposure Drafts and Standards submitted by the Standards Development Committee.
- 175. The Working Party believes that the proposals set out in paragraphs 172-174 will ensure reasonably widespread acceptance for IASC's work, without undue risk of paralysing the work of the Standards Development Committee. It will also give genuine decision-making power to both the Standards Development Committee and the Board.
- 176. The Working Party recommends that the Board should have the right to add projects to the Standards Development Committee's work plan, but should not have the right to remove projects from the work plan. The Standards Development Committee should seek regular guidance on its work plan from the Board. The Standards Development Committee should also discuss the main technical issues in all its proposals with the Board in depth at an early stage and as projects progress.
- 177. The Working Party considers that:
  - (a) a majority of 60% (i.e. 15 votes out of 25) should be required to consent to changes in the Constitution (see paragraph 183); and

- (b) all other Board decisions should require a simple majority of those present.
- 178. The Board would probably need to meet three times a year.

## **Trustees - Responsibilities**

- 179. The Working Party proposes that the Advisory Council should be replaced by Trustees with powers and responsibilities assigned by a revised IASC Constitution, rather than delegated to them by the Board. These powers and responsibilities would include:
  - (a) appointing members of the Standards Development Committee and the Standing Interpretations Committee, Board Members and the Chairmen and Vice-Chairmen of the Standards Development Committee, Board and Standing Interpretations Committee. The Trustees should draw up written criteria for this purpose;
  - (b) liaising with Board Members to ensure that Board Delegates are appointed who have the appropriate personal qualities to contribute to the Board and who, in aggregate, have an appropriate mix of backgrounds and experience. The Trustees would have the power to veto the appointment of any individual Board Delegate. The Working Party believes that the Trustees should exercise this power extremely rarely and proposes that such a veto should require the agreement of nine of the twelve Trustees. The Trustees should draw up written criteria for this purpose;
  - (c) ratifying the appointments (by the Chairman of the Standards Development Committee) of the Technical Director and Commercial Director;
  - (d) monitoring the ability of IASC's structure to lead to results that meet IASC's objectives;
  - (e) reviewing the effectiveness of the Standards Development Committee and the Board in meeting IASC's objectives;
  - (f) reviewing broad strategic issues and relationships with key constituents (for example, IOSCO, IFAC, the World Bank Group and national standard setters) in consultation with the Board and the Standards Development Committee;
  - (g) promoting IASC and its work;
  - (h) approving IASC's budget and monitoring the efficient use of its resources;
  - (i) fund-raising;
  - (j) publishing an annual written report on IASC's activities and the work of the Trustees and presenting a formal report to the members of IASC every two and a half years; and
  - (k) proposing changes to IASC's Constitution.
- 180. The Trustees should have the power to set up sub-committees as necessary. In particular, the Trustees should set up a Selection Sub-committee consisting of Trustees. The Selection Sub-committee would identify candidates for the Standards

Development Committee (in consultation with the Chairmen and Vice-Chairmen of the Board and Standards Development Committee and with national standard setters and other organisations who employ potential candidates), Standing Interpretations Committee, Board and "at large" Trustees.

- 181. The Trustees should do nothing to impair the independence and objectivity of the Board, the Standards Development Committee or the Standing Interpretations Committee and should not participate in their technical decisions. Thus, the Trustees should not intervene in technical disputes on individual projects between the Standards Development Committee and the Board. In the unlikely event of prolonged or repeated disputes between the Standards Development Committee and the Board, the Trustees would need to consider whether IASC's structure enables IASC to meet its objectives.
- 182. To avoid conflicts of interest, fund-raising should be the responsibility of the Trustees and not a responsibility of members of the Standards Development Committee and Board Delegates.
- 183. Changes to the Constitution should require approval by all of the following:
  - (a) 60% of the Trustees (8 votes out of 12);
  - (b) 60% of the Board (15 votes out of 25); and
  - (c) a simple majority of those members of IASC who vote on the proposed change.

## **Standing Interpretations Committee**

- 184. The Working Party believes that the Standing Interpretations Committee (SIC) should continue to exist as a separate body, because neither the Board nor the Standards Development Committee would have sufficient time to develop their own Interpretations.
- 185. At present, Board approval is required for a final Interpretation. Some members of the Working Party believe that this should continue. However, a majority of the Working Party would prefer final approval by the Standards Development Committee, to minimise delays in issuing Interpretations that may be needed urgently and because they believe that the SIC's own due process makes formal approval by the Board unnecessary. They propose that this should require the same majority as a decision to submit an Exposure Draft or Standard to the Board for approval (seven votes out of 11).
- 186. When the SIC submits final Interpretations to the Standards Development Committee for approval, it should also send a copy to Board Members so that they can comment to the Standards Development Committee before it approves the final Interpretation.
- 187. The Working Party believes that members of the SIC should be appointed by the Trustees. To avoid delays in filling vacancies on the SIC, the Working Party recommends that the Trustees should have the power to establish a Sub-Committee for this purpose.

- 188. The Working Party believes that one member of the Standards Development Committee and one Board Delegate should attend SIC meetings as observers.
- 189. It is too early to assess whether any other changes are needed to the structure or procedures of the Standing Interpretations Committee (SIC), which was formed in 1997.

# **Consultative Group**

- 190. The Working Party believes that IASC should maintain the Consultative Group as a useful forum for dialogue with organisations (mainly international) that have an interest in financial reporting. The Consultative Group should meet once or twice a year with the Chairmen of the Standards Development Committee, Board and Trustees and others as appropriate. The meetings should be chaired by the Chairman of the Board. The purpose of the meetings should be for the Consultative Group to:
  - (a) give feedback on IASC proposals, guidance on work plan priorities and advice on relationships with key constituencies; and
  - (b) receive reports on IASC's work

#### Members of IASC

- 191. The Working Party believes that the Members of IASC should continue to be the professional accountancy bodies that are members of IFAC and should continue to meet every two and a half years. The Members of IASC should meet under the Chairmanship of the Chairman of the Trustees to:
  - (a) receive a report by the Trustees on their activities over the preceding two and a half years;
  - (b) receive reports by the Chairmen of the Board and the Standards Development Committee on the activities, work programme and future strategy of the Board and the Standards Development Committee; and
  - (c) ratify (by a simple majority of those voting) any changes to IASC's Constitution that have been approved by the Trustees and Board (see paragraph 183).

## **Executive Committee**

192. The Working Party believes that the Board will no longer need an Executive Committee. Certain issues of an administrative nature would be handled conveniently by informal discussions between the Chairmen of the Standards Development Committee, Board and Trustees. As noted in Paragraph 180, the Trustees should have the power to set up sub-committees as they see fit.

#### Staff

193. As effective chief executive officer, the Chairman of the Standards Development Committee would take over the functions currently performed by the Secretary-General, except that the Chairman of the Board would carry out many of the representative duties as ambassador for IASC.

- 194. Under the Working Party's proposals, technical functions are the province of the Standards Development Committee and the Board, while commercial functions (including funding, copyright, office, equipment, communications) are the province of the Trustees. The Working Party proposes that:
  - (a) technical functions should be headed by a Technical Director, reporting to the Standards Development Committee; and
  - (b) commercial functions should be headed by a Commercial Director, reporting through the Chairman of the Standards Development Committee to the Trustees.

The Technical Director and Commercial Director should be appointed by the Chairman of the Standards Development Committee, subject to ratification by the Trustees.

- 195. To play an equal role in partnership with national standard setters, IASC needs a core of high-quality technical staff (at least eight), at a central location. Some projects would be joint projects with national standard setters and staffed predominantly by the national standard setter concerned. However, IASC's own staff would need to monitor the staff work on these projects to ensure that the output meets IASC's needs.
- 196. The Technical Staff would provide technical support to the Standards Development Committee and the Board. The Working Party does not believe that these functions need to be separated. The Commercial Director would be responsible for administrative support for the Board and the Trustees.

## **Legal Structure**

- 197. IASC is not a legal entity. This is administratively inconvenient. For example, it causes practical difficulty in enforcing IASC's copyright. Also, it creates a risk, however remote, that third parties might seek to make Members of IASC, Board Members, or even individual Board Delegates, liable for acts or omissions of IASC. The Working Party recommends that the Board should consider ways of establishing IASC as a legal entity. The Working Party believes that the establishment of a separate legal entity would also be more in keeping with IASC's future role.
- 198. The Working Party believes that IASC should investigate the possibility of seeking charitable or similar status in those countries where such status would assist fundraising.

#### **Timetable for Change**

- 199. A possible timetable for implementing the Working Party's proposals is set out in appendix 6. Certain aspects of the Working Party's proposals would require changes to IASC's Constitution. As explained above, constitutional changes must be discussed with the Council of IFAC and require a three-quarters majority of the IASC Board and approval by the Members of IASC as expressed by a simple majority of those voting. The next meeting of the Members of IASC is planned for May 2000.
- 200. The possible timetable in Appendix 6 assumes that IASC plans for these changes as soon as possible on the assumption that the members of IASC approve the changes in May 2000. For example, the process of selecting the Trustees should begin at an early

stage, so that provisional Trustees can, on a provisional basis, begin selecting members of the Board and Standards Development Committee. The provisional appointments could then come into effect as soon as the Members of IASC have approved the revised constitution. The revised structure would then come into effect from 1 July 2000, when the term of the current Board expires. The Working Party recognises that this is a challenging timetable, which relies on the optimistic assumption that a provisional selection process can be largely completed before the necessary constitutional changes have been approved.

## **DUE PROCESS**

201. To safeguard IASC's legitimacy, IASC's due process must ensure that International Accounting Standards are of high quality, requiring transparent and comparable information that will help participants in capital markets and others to make economic decisions, and that its standards are acceptable to the users and preparers of financial statements. Given IASC's higher profile today, the greatly increased interest in its work and IASC's changing role, it is appropriate to seek improvement of IASC's due process.

# **Open Meetings or Closed Meetings?**

- 202. All IASC meetings are closed, although the Board agreed in November 1998 that Board meetings should be open to the public from 1999. The possible advantages of closed meetings are that:
  - (a) participants may find it easier to engage in open exchanges of ideas when they do not feel that they are on public display. This may be particularly true for participants from cultures that do not encourage public dissent and for those whose mother tongue is not English;
  - (b) the debates may be less politicised than in open meetings; and
  - (c) open meetings may lead to delays and extra cost.
- 203. The possible advantages of open meetings (a "sunshine" policy), are that they would:
  - (a) lead to better standards (through more effective public scrutiny); and
  - (b) increase public confidence in IASC and strengthen its legitimacy.
- 204. The Working Party believes that:
  - (a) all formal discussions of the Standards Development Committee, Standing Interpretations Committee and Board on technical issues should be open to the public. However, certain discussions (primarily selection, appointment and other personnel issues) would need to be held in private. Portions of the Trustees' meetings should also be open to the public, at the discretion of the Trustees;
  - (b) IASC should make more use of new technology (such as the Internet, the web site, electronic observation of meetings), to overcome geographical barriers and the logistical problems in arranging for members of the public to attend open meetings of an international body;

- (c) IASC should publish in advance the agendas for each meeting of the Standards Development Committee, Standing Interpretations Committee, Board and Trustees and should publish promptly the decisions made at those meetings (IASC currently publishes the agenda for Board meetings in its quarterly newsletter, *Insight*, and on its web site. IASC also publishes Board decisions immediately after each Board meeting in *Update* and SIC decisions in *News From The SIC*); and
- (d) when IASC publishes a Standard, it should continue its recently adopted practice of publishing a Basis for Conclusions to explain publicly how it reached its conclusions and to give background information that may help users of IASC standards to apply them in practice. IASC should also publish dissentient opinions (IASC's current Constitution prohibits this). These would be brief summaries, prepared by IASC's staff, of the reasons given by members of the Standards Development Committee and Board for voting against the Standard.

## **Public Hearings and Field Tests**

- 205. The Working Party believes that the Standards Development Committee should make use of the following, although there should be no requirement to do so for every project:
  - (a) "public hearings" to discuss proposed standards (Steering Committees have occasionally held similar meetings in the past, for example, in 1995, on Segment Reporting, and in 1997, on Financial Instruments); and
  - (b) field tests (both in developed countries and in emerging markets) to ensure that proposals are practical and workable.

Where practicable, public hearings and field tests should be co-ordinated with national standard setters.

#### **Co-ordination with National Due Process**

- 206. An important objective of the Working Party's model is closer co-ordination between IASC's due process and the due process of national standard setters, who will necessarily remain autonomous. The Working Party recognises that many national standard setters will not give up their own due process, nor can they give an irrevocable undertaking that they will tie themselves completely on every project to IASC's due process. However, the Working Party believes that IASC should aim to integrate IASC's due process more closely with national due process. This is something that will probably not happen overnight but will occur gradually as the relationship between IASC and national standard setters evolves. The Working Party's desired outcome is the following procedure for most, and preferably all, projects that have international implications:
  - (a) IASC and national standard setters would co-ordinate their work plans so that when IASC starts a project, national standard setters would also add it to their own work plans so that they can play a full part in developing an international consensus. Similarly, where national standard setters start projects, IASC would consider whether it needs to develop a new Standard or revise its existing Standards;

- (b) IASC and national standard setters would co-ordinate their timetables so that national standard setters would aim to publish their own proposals at the same time as IASC proposals and so that the results from national exposure are available in time for IASC to consider, and vice versa. IASC may need, in certain cases, to slow down its own timetable to some extent so that national standard setters can satisfy their own due process requirements. However, to avoid giving national standard setters a veto in IASC's process, IASC would sometimes need to issue its own proposals without significant delay, even if some national standard setters were not yet ready to issue their own proposals;
- (c) members of the Standards Development Committee would not be required to vote for an IASC treatment in their national standard setters, since each country would remain free to adopt IASC standards with amendments or to adopt other standards. However, the existence of an international consensus is clearly one factor that members of national standard setters would consider when they decide how to vote on national standards;
- (d) IASC would continue to publish its own Exposure Drafts and other documents for public comment;
- (e) national standard setters would publish their own Exposure Drafts at approximately the same time as IASC Exposure Drafts and would seek specific comments on any significant divergences between the two Exposure Drafts. The Working Party expects that market forces would make such divergences increasingly rare. In some instances, national standard setters may ask for specific comments on issues of particular relevance to their country or include more detailed guidance than is included in an International Accounting Standard; and
- (f) national standard setters would follow their own full due process, which they would, ideally, choose to integrate with IASC's due process. Issues arising would be considered by national standard setters. This integration would avoid unnecessary delays in completing standards and would also minimise the likelihood of unnecessary differences between the various standards that result. In the same way as for Exposure Drafts, the Working Party expects that it will become increasingly rare for national standard setters to adopt standards that differ from International Accounting Standards.

## **Comment periods**

- 207. Until 1996, IASC allowed a comment period of up to six months for Exposure Drafts. In 1996, IASC accelerated its work programme for the core standards that must be completed before IOSCO will consider endorsing International Accounting Standards for cross-border capital raising and listing purposes in all global markets. Since 1996, the comment period for Exposure Drafts has been three months and several Exposure Drafts have been issued at short intervals. Some have expressed a concern that a comment period of three months may sometimes be too short for respondents to understand and evaluate proposals in sufficient depth, particularly for those respondents who rely on local translations of IASC proposals.
- 208. The acceleration of the work programme in 1996 certainly placed strains on IASC's constituents. However, most constituents accepted this, in the hope that it would lead to IOSCO endorsement. The Working Party believes that there is now a case for

IASC to extend its comment periods, although with modern communications it may no longer necessary to have comment periods as long as six months. A minimum of four months may be appropriate, although particularly complex or controversial issues may warrant longer comment periods.

# **Charging for Exposure Drafts**

209. Some have argued that IASC deters comments by charging for Exposure Drafts, Issues Papers and similar documents. The Working Party acknowledges that this may be the case. However, the need to encourage comments must be balanced against the general principle that those who benefit from IASC should pay for its work.

#### **Translations**

- 210. In the past, IASC's policy has been to publish International Accounting Standards (as well as other documents, such as Exposure Drafts) only in English. This policy creates a barrier to the use and understanding of International Accounting Standards, particularly in those countries that adopt the standards through domestic legislation.
- 211. The Board has sometimes given authority to the Members of IASC and others to prepare translations of the approved text of exposure drafts and International Accounting Standards. International Accounting Standards have been translated into more than 30 languages.
- 212. There have been criticisms of some existing translations of International Accounting Standards. Although IASC has no responsibility for these translations, these criticisms may reflect adversely on IASC. More importantly, lack of quality in translations may lead to divergent or even unacceptable interpretations of IASC's standards. The Working Party believes that IASC should explore ways of establishing quality control of these translations, possibly by working with local standard setters and accountancy bodies.
- 213. IASC has recently published German and Russian translations of its standards. These are the first translations that IASC has undertaken. IASC aims to publish further translations and has already started work on a Polish translation. The objective of such translations is to:
  - (a) promote the use of International Accounting Standards;
  - (b) ensure that users of International Accounting Standards have access to high quality translations; and
  - (c) raise revenue.
- 214. The Working Party recommends that IASC should publish translations of its standards in other languages, preferably on a self-financing basis. For certain languages, a translation of International Accounting Standards may not generate sufficient revenue to cover the cost. It may be possible to obtain funding for these translations from national and international development agencies and similar bodies.

BLANK PAGE

# 6. IMPLEMENTATION, ENFORCEMENT AND TRAINING

- 215. IASC's work does not end with the publication of standards. IASC must also work in partnership with national standard setters and regulators and encourage them to adopt requirements that are consistent with International Accounting Standards. The Working Party believes that it is not the role of IASC to review national standards in order to assess actively whether those national standards result in compliance with International Accounting Standards.
- 216. Some argue that IASC should be more pro-active in identifying departures by preparers from International Accounting Standards (or from national requirements that are consistent with International Accounting Standards) and reporting those departures to national enforcement agencies, IOSCO and the professional accountancy bodies that are members of IFAC and IASC. In principle, the Working Party believes that it would be desirable for IASC to play a more pro-active role in this area. However, the Working Party recognises that IASC does not have the resources to do this effectively. Also, IASC lacks both legal authority to take action and legal protection from those who dispute alleged departures. In the Working Party's view, identifying and dealing with such departures is primarily a matter for auditors, professional accountancy bodies, IFAC, national enforcement agencies and supranational bodies such as IOSCO and the Basle Committee.
- 217. The Working Party believes that IASC should give advice to national regulators and other enforcement agencies in their efforts to enforce national standards that comply with International Accounting Standards, but only if the agency in question both:
  - (a) pays for the advice on a fully self-financing basis; and
  - (b) gives IASC satisfactory indemnities against legal action by those who dispute alleged departures.
- 218. IFAC's draft strategic plan proposes that IFAC should:
  - (a) propose a policy and procedures for reviewing the commitments of member bodies in meeting IFAC and IASC standards and guidelines and for ensuring that member bodies have appropriate arrangements for monitoring the observance by their members of such standards and guidelines; and
  - (b) publish the first results of a review of the extent to which member bodies meet their commitments and of the appropriateness of their monitoring procedures.

IFAC's education committee may also have a role.

219. Some national standard setters publish training material, illustrative examples and other implementation guidance, such as staff bulletins. However, IASC has published such material only on an experimental basis, because of lack of human resources. Similarly, IASC does not generally give training courses and does not provide a technical enquiry service. This is because of concerns that such material, although not officially authoritative, might be perceived to have quasi-authoritative status - and so might appear to undermine IASC's due process. Nevertheless, such material would undoubtedly meet a need and it may also be a useful source of revenue.

# Implementation, Enforcement and Training

220. The Working Party believes that IASC should consider publishing training material, illustrative examples and other implementation guidance, such as staff bulletins and, perhaps, also giving training courses. However, at this stage IASC should not provide a technical enquiry service.

# 7. FUNDING

#### **IASC'S CURRENT FUNDING**

- 221. IASC's current income and expenditure is summarised in appendix 4.
- 222. Currently, a charge is made to Board Members to cover the budgeted excess of expenditure over other income. The same amount is charged to all Board members, except that the Council of IFAC may decide to reimburse wholly or in part the share charged to one or more Board Members. IFAC currently reimburses the cost of one Board seat and this subsidy is shared equally by India, Sri Lanka and Zimbabwe. In addition, IFAC contributes one ninth of the total amount charged to Board Members.
- 223. IASC direct costs are remarkably low, largely because many volunteers give their time free, and because IASC relies heavily on research already carried out by national standard setters. Also, some organisations and individuals do not charge their full travel and other costs to IASC. In addition, joint projects with national standard setters sometimes involve staff support by those national standard setters at no cost to IASC (other than direct travel costs).

#### 224. IASC reimburses:

- (a) the travelling, hotel and incidental expenses of attendance at Board meetings by one Board Representative from each Board Member;
- (b) the Chairman for expenses incurred in attending Board meetings and otherwise on behalf of IASC; and
- (c) the travelling, hotel and incidental expenses of members of Steering Committees, the Advisory Council and the Executive Committee.
- 225. IASC does not take out insurance cover for members of the Board or Steering Committees.

### FUNDING IASC FOR THE FUTURE

- 226. The current method of funding IASC is open to a number of criticisms:
  - (a) although international travel and the need to research issues in an international context suggest a need for substantial resources, IASC's resources are remarkably modest. This leads to disproportionate demands on the time of Board Representatives, Technical Advisors and the staff;
  - (b) there is a conflict between the promotion of IAS (especially in poorer countries) and the need to raise funds ("the user pays"). For example, IASC's policy of charging for Exposure Drafts may deter comments;
  - (c) existing sources of revenue could be threatened by increased use of the Internet and by the increasing availability of national standards that are identical to, or very closely based on, International Accounting Standards;
  - (d) donations may undermine IASC's actual or perceived independence;

- (e) there is a lack of geographical balance in corporate donations. As a result, it may appear that IASC gives more weight to the views of countries that provide a higher level of donations;
- (f) IASC relies on volatile and uncertain sources of funding. This inhibits long-term planning, diverts scarce staff time and makes it difficult to recruit permanent staff;
- (g) the direct and indirect cost of Board seats deters developing and emerging countries from applying to join the Board;
- (h) the limitations of IASC's resources have forced it to prioritise projects aimed primarily at the needs of developed capital markets and to pay less attention to identifying and meeting any specific needs of developing countries and of countries in East and Central Europe and in Asia that are in transition from centrally planned economies to market driven economies; and
- (i) many organisations (including accountancy bodies that are not on the Board, stock exchanges, governments and national standard setters and others) benefit from IASC's work but do not provide funding. Also, the fact that IASC's only members are professional accountancy bodies is an impediment to fund-raising in some countries.
- 227. National standard setters are funded in a variety of ways, for example:
  - (a) the Australian Accounting Standards Board raises half its operating costs from government contribution, one third from the two accounting bodies and the balance from sale of publications. A special surcharge on stock exchange filing fees has been agreed for a two-year term, to help fund a programme to harmonise Australian standards with International Accounting Standards;
  - (b) Canada's Accounting Standards Board is funded entirely by the Canadian Institute of Chartered Accountants (CICA), through membership fees and sales of courses and publications, although the CICA's Task Force on Standard Setting recently recommended that external funding sources should be pursued, with the goal of achieving a more broadly-financed accounting standards function:
  - (c) the UK's Accounting Standards Board raises about 85% of its operating costs in roughly equal contributions from national government, the business community (through a combination of stock exchange listing fees and funding from the Bank of England on behalf of the banking sector and a variety of other financial institutions) and the accounting profession. The balance of its budget comes from sale of publications; and
  - (d) the USA's Financial Accounting Standards Board raises two thirds of its operating costs from the sale of publications and the balance is raised by the Financial Accounting Foundation through contributions by accounting firms, industry, banks and investment firms.
- 228. The level of funding needed by IASC, and the appropriate balance between different sources of funding, will depend on the nature of any changes made to IASC's structure and activities.

- 229. Preferably, IASC would need to increase its annual funding from approximately £2 million to around £5 million at current prices to implement the possible revised structure and due process described in paragraphs 124-214 (see appendix 4 for a breakdown). This amount excludes costs that, under the Working Party's proposals, would be borne by national standard setters and others. These costs would amount to between, perhaps, £1 million and £1.5 million. The amount of £5 million also excludes time costs for volunteers.
- 230. The Working Party believes that IASC needs more secure funding based on a formula, not a constant money drive, so as not to divert Trustee and staff time. There may be a need to use different formulas in different countries. There are several different ways of raising national contributions:
  - (a) directly from Board members (including perhaps observers) and/or Members of IASC;
  - (b) from a variety of groups in each country who benefit from IASC's work (e.g. preparers, users, regulators, the accountancy profession); or
  - (c) indirectly from groups who benefit from IASC's work, with Trustees and/or Members of IASC in each country taking responsibility for collecting the contributions for their countries. For example, it may be desirable to have some degree of commitment to funding over some minimum period, perhaps through organisations in each larger country, to facilitate longer-term planning.
- 231. The Working Party supports the general principle that those who benefit from IASC's work should pay for its work. The beneficiaries include users of financial statements, business enterprises, auditors, the accountancy profession in general, stock exchanges, regulators, central banks, governments and other government and intergovernmental agencies. However, it is not easy to identify all of those who benefit from IASC's work or to devise a fair way of sharing the cost between the different groups of beneficiaries. The Working Party would welcome suggestions on this.
- 232. In looking at various funding models, the Working Party considered a number of points, including the following:
  - (a) the enterprises that gain the greatest financial benefit from IASC's work are listed enterprises. Therefore, stock exchanges should be an important source of funding. It would seem equitable that all stock exchanges should contribute on a collective basis to remove the incentive for some stock exchanges to be "free riders" benefiting from IASC's work without paying for it;
  - (b) Trustees or member bodies could, perhaps, act as agents for fund-raising in their own countries/constituencies, working to targets agreed to be fair;
  - (c) representation of a broad range of constituencies on the Standards Development Committee, Board and Foundation should help fund-raising, as constituents will be more willing to fund a process in which they have representation;
  - (d) IASC may be able to persuade enterprises to endow IASC with permanent capital, as a source of investment income to fund part or all of IASC's work;

- (e) without a fair and equitable agreement for sharing publications revenue, publications revenue might drop if national standard setters adopted IAS and companies looked to national requirements instead of to IASC pronouncements;
- (f) by co-operating on projects on a rotational basis, national standard setters may save substantial costs overall. Therefore, it would be reasonable for national standard setters to devote significant resources to the Standards Development Committee (salary of the individual serving on the Standards Development Committee, travel costs, staff support, space, communications);
- (g) professional accountancy bodies carry out the standard setting role in many countries. Some of them may be reluctant to finance an international body when the majority of their local members do not operate in the international arena, as those local members may not realise how international co-operation leverages the resources that are available for standard-setting. However, they may be able to contribute funding not only in cash but also through secondments and through outsourcing of work to them; and
- (h) developing countries are unlikely to have the resources to pay a full contribution towards the cost of IASC. However, although any contribution structure is likely to include a subsidy from more established economies, this is likely to be in the public interest and in the interests of both developing and more established economies. Bodies such as the United Nations and the International Finance Corporation may be willing to support translations of International Accounting Standards and other work by IASC for emerging markets. They may also be willing to fund part of the cost of a Board seat for developing countries, as IFAC does at present.
- 233. The Working Party sees merit in a funding model that relies more or less equally on funding from a number of reasonably well-defined groups. An example would be a model that looks to the accountancy profession, government and the business and financial community to provide roughly equal proportions of IASC's funding. The most effective and efficient way to collect the business community's contribution might be through stock exchanges. The Working Party recognises that funding is a vital issue and aims to develop a more detailed funding plan during the period for public comment on this Discussion Paper. In the meantime, the Working Party would welcome comments on funding.
- 234. IASC will need continued help with fund-raising from the Advisory Council until any new structure can be put in place.
- 235. The Working Party proposes that the Standards Development Advisory Committee should be run on a self-financing basis.

# 8. CONCLUSION

236. The completion of IASC's current work programme to develop the IOSCO core standards is now imminent. However, IASC will face even greater challenges as it works, in partnership with national standard setters, for further convergence of national accounting standards and practices with high-quality global accounting standards. Therefore, it is vital to give IASC the right structure for the beginning of the twenty-first century. The Working Party invites all parties affected by accounting standards to play a full part in this important debate.

**BLANK PAGE**